Time to Event Analysis of Arthroplasty Registry Data

Marianne Knarberg Hansen Gillam MBBS, MBiostats

Thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy, January 2013

Discipline of Public Health School of Population Health Faculty of Health Sciences The University of Adelaide Australia

Table of Contents

Tab	le of	Conten	S	i
List	of T	ables		V
List	of F	igures	vi	i
ABS	STRA	4СТ	iz	K
Dec	larat	ion	Х	i
Maı	nusci	ripts Co	ntributing to this Thesisxi	i
Pres	senta	tions A	ising from this Thesisxii	i
Ack	now	ledgmer	tsxv	V
Abb	orevi	ations	XV	i
1	Intr	oductio	1	1
	1.1	Backgr	bund	1
		1.1.1	Arthroplasty registries	2
		1.1.2	Time to event analysis	2
		1.1.3	Regression models	1
		1.1.4	Arthroplasty histories	5
	1.2	Thesis	im	5
	1.3	Thesis	outline	7
2	Ana	lysis of	time to event data	9
	2.1	Non-pa	rametric methods	1
	2.2	The Co	x proportional hazards model1	1
		2.2.1	Time-dependent covariates13	3
		2.2.2	Time- varying coefficients12	3
	2.3	The ad	litive Aalen model14	1

	2.4	Multiple	events	15
		2.4.1	Multi-state models	16
		2.4.2	Competing risks	18
		2.4.3	Regression models for competing risks	20
	2.5	Analysis	of joint replacement registry data	22
3	Dat	a sources.		24
	3.1	The Aust	ralian Orthopaedic Association National Joint Replacement	
		Registry		24
	3.2	The Norv	wegian Arthroplasty Register	26
	3.3	Ethical co	onsiderations	26
4	Cor	npeting ri	sks survival analysis applied to data from the Australian	
	Ort	hopaedic	Association National Joint Replacement Registry	28
	4.1	Preface		28
	4.2	Statemen	t of Authorship	29
	4.3	Article		31
		4.3.1	Abstract	31
		4.3.2	Introduction	32
		4.3.3	Materials and methods	34
		4.3.4	Results	35
		4.3.5	Discussion	44
	4.4	Addition	al Discussion	48
5	Diff	erent con	peting risks models applied to data from the Australian	
	Ort	hopaedic	Association National Joint Replacement Registry	51
	5.1	Preface		51
	5.2	Statemen	t of authorship	52
	5.3	Article		53
		5.3.1	Abstract	53

		5.3.2	Introduction	54
		5.3.3	Background to statistical methods	55
		5.3.4	Patients and methods	
		5.3.5	Results	60
		5.3.6	Discussion	67
	5.4	Addition	al discussion	71
6	Mu	lti-state n	nodels and arthroplasty histories after unilateral total	hip
	artl	hroplastie	es. Introducing the Summary Notation for Arthroplast	у
	His	tories		75
	6.1	Preface		75
	6.2	Statemer	nt of Authorship	76
	6.3	Article		77
		6.3.1	Abstract	77
		6.3.2	Introduction	78
		6.3.3	Material and methods	79
		6.3.4	Results	
		6.3.5	Discussion	
	6.4	Supplem	entary article	92
7	The	e progress	sion of end-stage osteoarthritis: Analysis of data from	the
	Aus	stralian a	nd Norwegian joint replacement registries using a mu	ti-state
	mo	del		
	7.1	Preface		
	7.2	Statemer	nt of Authorship	
	7.3	Article		101
		7.3.1	Abstract	101
		7.3.2	Introduction	102
		7.3.3	Material and methods	103
		7.3.4	Results	107

		7.3.5	Discussion	116
8	Sun	nmary ai	nd conclusions	120
	8.1	Main fi	ndings and contributions	120
		8.1.1	Non-parametric competing risks methods and arthroplasty	1
			data	120
		8.1.2	Competing risks regression and arthroplasty data	121
		8.1.3	Multi-state models and arthroplasty histories	122
		8.1.4	Application of multi-state models and osteoarthritis	123
	8.2	Limitati	ons and future directions	124
	8.3	Conclus	sion	126
9	Ref	erences.		127

List of Tables

Table 4.1: Distribution of outcomes for the three study groups.	36
Table 4.2: Percent estimates (95% confidence interval) of revision in patients aged 75-84 years with FNOF.	l 38
Table 4.3: Percent estimates (95% confidence interval) of revision in patients with FNOF receiving Austin-Moore or Thompson prostheses.	42
Table 4.4: Percent estimates (95% confidence interval) of revision in patients with osteoarthritis who underwent total hip replacement – by age group	44
Table 4.5: Data from the Norwegian Arthroplasty Register. Percent estimates (with 95% confidence interval) of revision in patients with osteoarthritis who underwent total hip replacement – by age group.	1 5 49
Table 5.1: Distribution of outcomes by covariate status	60
Table 5.2: Estimates of hazard and subdistribution hazard ratios of revision based of a Cox-Aalen and a modified Fine and Gray model respectively, effect of fixation varies with time	on of 63
Table 5.3: Relationship between HRs and subHRs, modified from Lau et al. [109]	73
Table 5.4: Cause specific hazard ratios (HRs) and subdistribution hazard ratios (subHRs) for different covariate for a stratified Cox PH model and a modified Fine and Gray model respectively.	74
Table 6.1: Numbers and percent of events in the 10 state model (see Figure 6.1) at the end of the study period for patients whose first arthroplasty was a either a left or a right total hip arthroplasty for osteoarthritis.	84
Table 6.2: Effect of sex adjusted for age on the transition hazards between states (s Figure 6.1) for patients whose first arthroplasty was a total hip arthroplasty for osteoarthritis.	see 88
Table 7.1: Distribution of individuals according to covariates. 1	07
Table 7.2: Numbers and percent of events in the multi state model (Figure 7.1) at t end of the study period for patients whose first arthroplasty was a eithe a hip or a knee arthroplasty for OA	he er 08

Table 7.3: Effect of side of first arthroplasty (hip or knee) on hazards for selected	ed
transitions in the model	111

List of Figures

Figure 2.1:	Traditional survival model with one event of interest and hazard rate $\lambda(t)$
Figure 2.2:	Competing risks model with two absorbing states and cause specific hazards $\lambda_1(t)$ and $\lambda_2(t)$
Figure 2.3:	Multi-state model with three transient states, one absorbing state (dead) and transition intensities $\lambda_{gh}(t)$ from state g to state h, where $g = 0, 1, 2$ and $h = 1, 2, 3$
Figure 4.1:	Estimates of revision by type of prosthesis in patients with FNOF aged 75-84 years
Figure 4.2:	Estimates of death by type of prosthesis in patients with FNOF aged 75- 84 years
Figure 4.3:	Relative overestimation of KM estimates compared to CIF estimates by years after primary procedure and type of prosthesis; patients aged 75-84 years with FNOF
Figure 4.4:	Estimates of revision by type of prosthesis (cementless Austin Moore vs. cemented Thompson) in patients with FNOF
Figure 4.5:	Estimates of death by type of prosthesis (cementless Austin Moore vs. cemented Thompson) in patients with FNOF
Figure 4.6:	Estimates of revision by age group in patients with OA and THA43
Figure 4.7:	Estimates of death by age group in patients with OA and THA
Figure 4.8:	Estimates of revision by age group in patients with OA and THA (data from the NAR)
Figure 4.9:	Estimates of death by age group in patients with OA and THA (data from the NAR)
Figure 5.1:	Estimates of CIFs for revision for each variable
Figure 5.2:	Estimates of CIFs for death for each variable

Figure 5.3:	Effect of cementless fixation vs. cemented fixation on the subdistribution hazard of revision with 95% point wise confidence bands. The slope of the curve indicates the additional probability of revision for cementless fixation in relation to cemented fixation
Figure 5.4:	Comparison of predictions of revision based on Cox-Aalen (grey) and modified Fine and Gray models (black); effect of type of fixation varies with time
Figure 6.1:	Multi-state model with 10 states for patient who received a first hip arthroplasty followed by possibly a second arthroplasty (hip or knee), revisions of these, and death
Figure 6.2:	Example of the multi-state model with SNAH code on a subsample of patients who received a left hip prosthesis as first arthroplasty, followed by another primary arthroplasty or a revision of the left hip. (Number of events in parentheses)
Figure 6.3:	State occupation probabilities for patients in 3 age groups after first hip arthroplasty based on the model in Figure 6.1 (revision: state 2, hip: state 3, knee: state 4, dead: state 10, other: state 5-9)
Figure 7.1:	Multi-state model
Figure 7.2:	Comparing hazards of receiving a left knee arthroplasty between individuals who had received a right hip arthroplasty with individuals who had received a left hip arthroplasty. HR: hazard ratio, $\lambda(t R)_{1\rightarrow3}$: hazard of receiving a left knee given that first hip was a right hip, $\lambda(t L)_{1\rightarrow3}$: hazard of receiving a left knee given that first hip was a left hip
Figure 7.3:	Comparing hazards of receiving a right knee between individuals who had received arthroplasties in right hip and left knee with individuals who had received arthroplasties in left hip and left knee. HR: hazard ratio, $\lambda(t R)_{3\to8}$: hazard of receiving a right knee given that first hip was a right hip, $\lambda(t L)_{3\to8}$: hazard of receiving a right knee given that first hip was a left hip
Figure 7.4:	Estimated probabilities for receiving a knee arthroplasty after having received a hip arthroplasty (AU: Australia, NOR: Norway, left panel: state 3, right panel: state 4)
Figure 7.5:	Estimated probabilities for receiving a hip arthroplasty after having received a knee arthroplasty (AU: Australia, NOR: Norway, left panel: state 3, right panel: state 4)

ABSTRACT

Background: Arthroplasty registry data are traditionally analysed using standard survival methods, that is, Kaplan-Meier survival curves and the Cox proportional hazards model. The outcome of interest is usually the time from the primary procedure until occurrence of a single event – revision of the prosthesis. Other outcomes may also be of interest, for example, time to death, time to receiving another arthroplasty and the association between covariates and these events. The rise in life expectancy of the population combined with an increasing number of joint replacements being performed has resulted in many patients experiencing several joint replacement procedures during their lifetime. The analyses of registry data such as these require the use of more sophisticated statistical methods. Application and evaluation of statistical methods to analyse registry data containing complex arthroplasty histories are lacking.

Aim: The aim of this thesis was to investigate the use of statistical methods in the analysis of multiple event data contained in arthroplasty registries. Within this broad aim the objectives were to investigate the use of competing risks methods in estimating the risk and rate of revision, investigate methods for handling covariates with time-varying effect, investigate the use of multi-state modelling techniques in providing a more comprehensive analysis and description of complex arthroplasty histories than traditional survival methods and to develop a notation system to facilitate the description and analysis of arthroplasty event history data.

Methods: Data were obtained from the Australian Orthopaedic Association National Joint Replacement Registry and the Norwegian Arthroplasty Register. Estimates of revision from the Kaplan-Meier method were compared to estimates from the cumulative incidence function which accounts for the competing risk of death. Effects of covariates on the rate and risk of revision were estimated with competing risk regression and compared to estimates from the Cox proportional hazards model. Multi-state models were set up and applied to the data. The Summary Notation for Arthroplasty Histories (SNAH) was developed in order to help manage and analyse this type of data.

Results: The Kaplan-Meier method substantially overestimated the risk of revision compared to estimates using competing risks methods when the incidence of the competing risk of death was high. The influence of some covariates on the hazard rate was different to the influence on the actual probability of occurrence of the event as this was modulated by their relationship with the competing event. Multi-state models, in combination with SNAH codes, were well suited to the management and analysis of arthroplasty registry data on patients who had multiple joint procedures over time. Multi-state modelling techniques proved useful in the investigation of the progression of end-stage osteoarthritis in data from two national arthroplasty registries.

Conclusion: In the presence of competing risks, the Kaplan-Meier method may lead to biased estimates of the risk of revision, and hazard ratios obtained from the Cox proportional hazards model and competing risks regression models need to be interpreted with care. Multi-state models provide a useful tool to analyse data containing complex arthroplasty histories.

Declaration

I certify that this work contains no material which has been accepted for the award of any other degree or diploma in any university or other tertiary institution and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made in the text. In addition, I certify that no part of this work will, in the future, be used in a submission for any other degree or diploma in any university or other tertiary institution without the prior approval of the University of Adelaide and where applicable, any partner institution responsible for the joint-award of this degree.

I give consent to this copy of my thesis when deposited in the University Library, being made available for loan and photocopying, subject to the provisions of the Copyright Act 1968.

The author acknowledges that copyright of published works contained within this thesis resides with the copyright holder(s) of those works.

I also give permission for the digital version of my thesis to be made available on the web, via the University's digital research repository, the Library catalogue and also through web search engines, unless permission has been granted by the University to restrict access for a period of time

Signed:

Marianne KH Gillam (Candidate)

Date:

Manuscripts Contributing to this Thesis

Gillam MH, Ryan P, Graves SE, Miller LN, de Steiger RN, Salter A. Competing risks survival analysis applied to data from the Australian Orthopaedic Association National Joint Replacement Registry. *Acta Orthopaedica* 2010;81(5):548-55. DOI: 10.3109/17453674.2010.524594

Gillam MH, Salter A, Ryan P, Graves SE. Different competing risks models applied to data from the Australian Orthopaedic Association National Joint Replacement Registry. *Acta Orthopaedica* 2011;82(5):513-20. DOI: 10.3109/17453674.2011.618918

Gillam MH, Ryan P, Salter A, Graves SE. Multi-state models and arthroplasty histories after unilateral total hip arthroplasties. Introducing the Summary Notation for Arthroplasty Histories. *Acta Orthopaedica* 2012;83(3):220-6. DOI: <u>10.3109/17453674.2012.684140</u>

Supplementary article data: http://www.actaorthop.org/sup_files/5260_SAD.pdf

Gillam MH, Lie SA, Salter A, Furnes O, Graves SE, Havelin LI, Ryan P. The progression of end-stage osteoarthritis: analysis of data from the Australian and Norwegian joint replacement registries using a multi-state model. *Osteoarthritis and Cartilage* (2012). First published online 26. December 2012. http://dx.doi.org/10.1016/j.joca.2012.12.008

Presentations Arising from this Thesis

Gillam MH. Competing Risks Survival Analysis Applied to Data from the Australian Orthopaedic Association National Joint Replacement Registry. Australasian Faculty of Public Health Medicine (AFPHM) annual meeting. Adelaide, November 2012.

Gillam MH, Lie SA, Salter A, Furnes O, Graves SE, Havelin LI, Ryan P. Investigation of the progression of end-stage osteoarthritis using data from the Australian and Norwegian joint replacement registries. Population Health Congress. "Population Health in a Changing World". Adelaide, 9-12 September 2012.

Gillam MH. Investigation of the progression of end stage osteoarthritis using data from the Australian and Norwegian joint replacement registries. School of Population Health Seminar Series. Adelaide, August 2012.

Gillam MH. Multi-state models and arthroplasty histories. AFPHM annual meeting. Adelaide, November 2011.

Gillam MH. Competing Risks Survival Analysis applied to data from the Australian Orthopaedic Association National Joint Replacement Registry. (Invited seminar) http://www.med.uio.no/imb/english/research/news-and-events/events/guest-lectures-seminars/2011/bmms-gillam.html. Seminar at Institute of Basic Medical Sciences, Faculty of Medicine, University of Oslo. Norway, October 2011.

Gillam MH, Lie SA, Salter A, Furnes O, Graves SE, Havelin LI, Ryan P. Studie av progresjon av coxartrose og gonartrose på data fra de australske og norske leddproteseregistrene (Study of progression of hip and knee osteoarthritis on data from the Australian and Norwegian joint replacment registries). Norwegian Orthopaedic Congress 2011. Oslo, Norway, 26-28 October 2011.

Gillam MH, Salter A, Ryan P, Graves SE. Regression Models for Competing Risks Applied to Data on Patients with Fractured Neck of Femur from the Australian Orthopaedic Association National Joint Replacement Registry (AOA NJRR). Faculty of Health Sciences 2011 Postgraduate Research Conference. Adelaide, August 2011.

Gillam MH. Multi-state models and arthroplasty histories. School of Population Health and Clinical Practice Seminar Series. Adelaide, August 2011.

Gillam MH, Ryan P, Graves SE, Miller LN, de Steiger RN, Salter A. Competing Risks Survival Analysis applied to data from the AOA National Joint Replacement Registry. The 70th Annual Scientific Meeting of the Australian Orthopaedic Association. Adelaide, 10-14 October 2010.

Gillam MH, Ryan P, Graves SE, Miller LN, de Steiger RN, Salter A. Competing Risks Analysis Applied to Joint Replacement Registry Data. Faculty of Health Sciences 2010 Postgraduate Research Conference. Adelaide, September 2010.

Gillam MH. Competing risks survival analysis applied to data from the AOA National Joint Replacement Registry. School of Population Health and Clinical Practice Seminar Series. Adelaide, August 2010.

Gillam MH. Statistical methods for survivorship analysis applied to arthroplasty registry data. School of Population Health and Clinical Practice Seminar Series. Adelaide, September 2009.

Acknowledgments

I would like to sincerely thank my supervisors, Phil Ryan and Amy Salter. This thesis would not have been possible without them. I have benefited from Phil's wisdom and linguistic precision, Amy's encouragement and demands for clarity, and their ongoing guidance and support. Thank you both.

Thank you to my co-authors from the Australian Orthopaedic Association National Joint Replacement Registry (AOA NJRR): Stephen Graves, who has been unfailingly generous with his advice and expertise, and to Richard de Steiger and Lisa Miller for their constructive comments. I am grateful for having had the opportunity to collaborate with Stein Atle Lie, Ove Furnes and Leif Havelin from the Norwegian Arthroplasty Register (NAR). They have provided invaluable comments and contributions, and it has been a real pleasure to work with them in my native language.

Thank you to the AOA, the NAR, the hospitals, the orthopaedic surgeons and their patients whose data made this work possible.

I am also thankful to the Faculty of Health Sciences Research Committee who awarded me a Postgraduate Travelling Fellowship for health science research and to the AOA NJRR for funding to attend a conference.

Thank you to my fellow PhD students, especially Allison, Aris, Chuangzhou, Helene, Nasreen, Vanda and Vicki in my thesis writing group, and to George and Lisa, my office colleagues, for providing support, inspiration, constructive feedback and motivation.

I am grateful to my parents, Lise and Egill, and my brother, Morten for their inspiration, great understanding and for always taking a keen interest in what I am doing.

Finally, thank you to my loving husband Charles and my three wonderful children Astrid, Magnus and Marit with whom I am so lucky to share my life.

Abbreviations

- AOANJRR: Australian Orthopaedic Association National Joint Replacement Registry
- CIF: Cumulative Incidence Function
- Cox PH: Cox Proportional Hazards
- CPR: Cumulative Percent Revision
- FNOF: Fractured Neck of Femur
- HR: Hazard Ratio
- KM: Kaplan-Meier
- NAR: Norwegian Arthroplasty Register
- OA: Osteoarthritis
- RD: Relative Differences
- SNAH: Summary Notation for Arthroplasty Histories
- SubHR: Subdistribution Hazard
- THA: Total Hip Arthroplasty
- TKA: Total Knee Arthroplasty

1 INTRODUCTION

This thesis investigates methods for analysis of time to event data in arthroplasty registry data. Arthroplasties—specifically joint replacements—are common surgical procedures. Registries collect and record data on these procedures and serve as important sources of data for evaluating specified outcomes of joint replacements. In order for arthroplasty registries to provide accurate information for their stakeholders, the use of appropriate statistical methods to analyse the data is of great importance. This thesis argues for the use of methods that appropriately account for the potential of multiple events, rather than standard univariate survival methods currently employed, when analysing time to event data contained in arthroplasty registries.

1.1 Background

Patient registries are organised systems that contain observational study data for populations characterised by particular diseases, conditions or exposures. They serve predetermined purposes such as monitoring safety of medical devices, measuring quality of care, describing the natural history of diseases, and determining health outcomes and treatment effects [1].

Patient registries are important sources of data for observational studies which are the mainstay of population health research. Traditionally, observational studies have had a lower ranking on the evidence scale than randomized controlled trials (RCTs) but the advantages of observational studies based on registry data are increasingly being recognized [2]. Some of these are that the data often come from large and varied groups of patients and the outcomes can therefore be generalized to a wide range of patients. They can be used pragmatically to evaluate treatments that are administered in the real world rather than those administered under what are often rigid RCT regimes and restrictive patient exclusion criteria. For rare diseases observational studies based on registry data can be used to measure outcomes where clinical studies are not feasible or

in situations where it is unethical to conduct RCTs. Furthermore, RCTs are often time consuming and resource intensive [3-6].

1.1.1 Arthroplasty registries

Patient registry data are particularly useful for evaluating outcomes of procedures which require relatively long follow up, such as the outcome of joint replacements [7, 8]. Sweden was the first country to establish a national joint replacement registry in 1975 [9], and following its success, several other countries have established national joint replacement registries [10]. One important purpose of these registries is to identify poorly performing prostheses thereby improving the quality of joint replacements [11]. In most countries prostheses are introduced on the market without prior clinical trials and thus continuous post marketing surveillance is important. Information on the collected data is typically disseminated in annual reports.

The incidence of joint replacements has been increasing in most countries over the past decade and it is expected that this will continue [12] as the incidence of the main risk factors such as osteoarthritis, age and obesity increase in the population. In the United States increases in primary hip and knee replacements of 174% and 673% respectively have been predicted for the period 2005 to 2030 [12]. In Australia there was an increase in primary total hip replacements and primary total knee replacements of 44% and 84% respectively from 2003 to 2012 [13].

1.1.2 Time to event analysis

In order to evaluate the outcome of joint replacement correctly, the use of appropriate statistical methods is important. Survival analysis, or time to event analysis, is commonly used in the analysis of registry data. The main outcome of interest in joint replacement registry data is 'time to revision' – the time interval between the date of

insertion of the prosthesis and the date of revision. Revision is a procedure where one or more of the prosthesis' components is removed and/or replaced. It signals failure of the prosthesis. A characteristic of time to event data¹ is that it may contain observations with incomplete information in that the exact survival time is not known, for example when a study ends before a revision is observed. These incomplete observations are called censored observations. The survival methods most commonly used in the analysis of joint replacement registry data are Kaplan-Meier survival curves [14] and Cox proportional hazards models [15]. These methods are extremely popular and the two papers introducing them are among the most cited statistical papers in the research literature [16]. The Kaplan-Meier method estimates a survival curve which indicates the proportion of subjects who have not yet experienced the event of interest at various time points. The method was initially based on actuarial life tables to estimate the proportion of subjects who had not yet died, but would eventually do so. The Kaplan-Meier method accounts for censored observations and thereby utilizes the information contained in the incomplete data. A crucial assumption is that censoring times and survival times are independent, that is, at time t patients whose time is censored will have the same risk of experiencing the event as those whose time is not censored, i.e. that censoring is non-informative. If patients are censored administratively this might be reasonable, but if they are lost to follow-up or have experienced another event then the assumption is often violated. For example, when estimating the risk of revision, subjects who have died (and therefore would be censored in the Kaplan-Meier method) clearly do not have the same risk of revision as subjects who are still alive, hence the censoring time is not independent. This violation of independence occurs due to a competing risk (where in this case the competing risk is death). Competing risks, where relevant, are

¹ Treatment of the word "data" as singular or plural has been a vexed question in both scientific and nonscientific literature for almost a century. In this thesis I will generally treat the word as plural but when it is either obviously being used as a "mass noun" or it would otherwise sound peculiar, I will take the liberty of treating it as singular.

often not accounted for in statistical analyses of survival data, leading to potential biased estimates of the survival curves [17]. Understanding the role of competing risks in registry data is one of the objectives of this thesis.

Another important function in survival analysis is the hazard rate, which is the instantaneous rate of occurrence of an event, conditional on the event having not yet occurred. The Cox proportional hazards (PH) model enables estimation of the effect of covariates on the hazard rate. The model is regarded as a semi-parametric because the baseline hazard function is not explicitly estimated. The covariates act multiplicatively on the baseline hazard rate and the effects are expressed as ratios. One key assumption of the model is that the hazard ratio is constant, that is, independent of time. The assumption of proportionality is not always plausible and is often assumed without justification [18]. Causes of non-proportionality are omission of covariates, incorrect functional form of the covariate or the existence of time dependent covariates or covariates with time-varying effects. Time dependent covariates are covariates that change value over time, whereas covariates with time-varying effect are covariates whose effect on the hazard rate changes over time. Depending on the cause, different methods exist to deal with non-proportional hazards in the Cox model [19], but no consensus on which method is the best appears to have been reached. An alternative model to the Cox PH model is Aalen's additive hazard model [20]. In Aalen's additive model the effect of covariates acts additively on the baseline hazard rate and both timedependent covariates and time-varying coefficients can easily be incorporated. The use of Aalen's model on registry data and exploration of methods for handling covariates with time-varying effects are also objectives of this thesis.

1.1.3 Regression models

In traditional survival analysis where the outcome is time to a single event, there is a one-to-one relationship between the hazard rate and the survival function. As a result, estimates from the Cox model can be used to directly estimate the effect of covariates on survival probabilities. However, in scenarios with competing risks this relationship is

more complicated because the probability of survival for one cause depends on the hazard rates for all the causes (cause specific hazards). One may find that some covariates have different effects on the cause specific hazard rates than they do on the survival probabilities. Hence, in the presence of competing risks the estimates from the Cox PH model require a different interpretation. Fine and Gray [21] have developed a method for competing risks scenarios to model the direct effect of covariates on survival probabilities for the event of interest. Regression methods for competing risks scenarios have not been used much in orthopaedic research, although they are often relevant. An objective of this thesis is to explore the use of competing risks regression models on arthroplasty registry data.

1.1.4 Arthroplasty histories

An increasing number of individuals have several joint replacement procedures and, with time, joint replacement registries will contain increasing amounts of data on individuals with complex arthroplasty histories. These histories may contain information on primary arthroplasties in multiple joints with associated revisions and rerevisions. One issue that arises as a result of this expansion is the management and handling of large datasets that evolve over time and contain multiple events for each individual. Developing a notation system for recording and communicating patient-level arthroplasty histories is one of the objectives of this thesis. Another issue is that statistical methods other than those traditionally employed for single-outcome data are required. Several methods exist to deal with time to event data with multiple outcomes. One method is to represent longitudinal data with several events as a multi-state model where an individual occupies and moves between a finite number of states characterised by conditions such as having had a joint replacement or a revision. A multi-state model can account for time dependent covariates which can be included as transient states. The competing risks model is an example of a simple multi-state model with two or more mutually exclusive end-states. One of the objectives of this thesis is to investigate the use and suitability of multi-state modelling techniques in providing a comprehensive analysis and description of complex arthroplasty histories contained in joint replacement registries. Furthermore, since joint replacement is most often performed for treatment of end-stage osteoarthritis, an objective is also to explore the use of multistate models on arthroplasty registry data to provide information on the progression of osteoarthritis.

1.2 Thesis aim

The overall aim of this thesis is to investigate the use of statistical methods in the analysis of time to event data in arthroplasty registries allowing for the potential of multiple events. Within this broad aim, the objectives are to:

- Apply competing risks methods to arthroplasty registry data to investigate the degree to which the probability of revision is biased using the Kaplan-Meier estimates and if present, to comment on whether this bias is clinically relevant.
- Apply competing risks regression models to determine the effect of covariates on the rate and the risk of revision and death.
- Investigate methods of handling covariates with a time-varying effect.
- Investigate the use of multi-state modelling techniques in providing a more comprehensive analysis and description of complex arthroplasty histories held in arthroplasty registries than traditional survival methods.
- Develop a notation system to facilitate the description and analysis of arthroplasty event history data.
- Apply multi-state models to investigate the progression of joint replacements (a surrogate for end-stage osteoarthritis) in data from two national joint replacement registries.

1.3 Thesis outline

Chapter 2 provides an introduction to time to event analysis, the important functions and issues with estimating the survival/failure curves and effects of covariates on the hazard rate. Multi-state models and competing risks are described. Literature pertaining to the specific aims (as outlined above) is reviewed and specific issues relating to the analysis of arthroplasty registry data are considered.

Chapter 3 describes the data sources in this thesis, the Australian Orthopaedic Association National Joint Replacement Registry (AOA NJRR) and the Norwegian Arthroplasty Register (NAR).

Chapters 4 to 7 consist of articles, three of which have been published and one which has been accepted for publication², each addressing different aspects of time to event analysis of arthroplasty registry data with multiple events. Chapter 4 contains a published article in which non-parametric competing risks methodology is applied to arthroplasty registry data and the results are compared to those obtained from the Kaplan-Meier method. In Chapter 5 the focus is on regression methods in competing risks scenarios and the inclusion of covariates with time-varying effects. The published article in this chapter investigates the effect of covariates on the rate and on the risk of revision.

Chapters 6 and 7 consider the use of multi-state models in the analysis of multiple events in joint replacement registry data. In Chapter 6, which contains a published article, the theory and issues with multi-state models are addressed and a multi-state model is developed. The Summary Notation for Arthroplasty Histories (SNAH) which facilitates the description and analysis of arthroplasty registry data is introduced. In Chapter 7 multi-state models are applied to data from the AOA NJRR and the NAR to

² First published online 26th December 2012.

investigate the progression of hip and knee osteoarthritis. The article in this chapter has been accepted for publication.

Chapter 8 contains a summary of the findings and contributions to knowledge in the thesis, addresses limitations and outlines topics for further research in this area.

2 ANALYSIS OF TIME TO EVENT DATA

This chapter begins by describing the basic elements of survival analysis and then goes on to discuss some of the approaches used to model both single and multiple event (event history) survival data. The aim of the chapter is to review literature on survival analysis of relevance to the aims of this thesis and to provide a convenient summary orientation to many of the modern statistical approaches to modelling survival data, some of which will be discussed in more detail in succeeding chapters.

Survival analysis – or analysis of time to event – is the analysis of data where the outcome is the time to the occurrence of one or multiple event(s) of interest. A particular characteristic of these data is that they may contain incomplete observations if events are not observed within the study period. These incomplete observations are referred to as censored events. There are several forms of censoring and each requires the use of different statistical methods in its analysis. Right censoring occurs if it is known that a survival time is greater than that observed, for example if an individual is lost to follow up or a study is closed for administrative reasons before the event has occurred. Left censoring occurs if a subject has experienced the event before the beginning of the study, but the exact timing of this event is not known. A related term is left-truncation which relates to the design of the study and means that subjects have been at risk of the event before the start of the study. Interval censoring takes place when the event occurs during the study period but the exact time is not known, only that it occurred in a certain time interval. Right censoring is the most commonly occurring type and will be considered in the following discussion.

The risk set at a particular point in time consists of individuals who (at time t) are at risk of experiencing the event of interest, that is, individuals who have entered the study and have survived up to time t. Individuals leave the risk set either when they experience the event, or when they are censored.

The main functions of interest in this context are *the survival function, the hazard function and the cumulative hazard function*. With T a random variable of survival times, the survival function is the probability of not experiencing the event by time t:

$$S(t) = P(T > t) = 1 - P(T \le t) = 1 - F(t),$$

where F(t) is the cumulative distribution function. The hazard function is usually interpreted as the instantaneous rate of experiencing an event (given that the individual is at risk at the beginning of a short time interval):

$$\lambda(t) = \lim_{\Delta t \to 0} \frac{1}{\Delta t} P(t \le T < t + \Delta t | T \ge t)$$
$$= -\frac{d \log\{S(t)\}}{dt},$$

where Δt denotes a small interval of time and *P* is the conditional probability that the survival time of a subject is between *t* and $t + \Delta t$, given that a subject has survived up to time *t*.

The cumulative hazard function represents the total accumulated risk in the interval (0, t):

$$\Lambda(t) = \int_{s=0}^{s=t} \lambda(s) ds.$$

The survival function, the hazard, and the cumulative hazard function are mathematically related such that:

$$S(t) = \exp\left(-\int_{s=0}^{s=t} \lambda(s) ds\right) = \exp(-\Lambda(t))$$

and

$$F(t)=1-\exp\left(-\int_{s=0}^{s=t}\lambda(s)ds\right).$$

Hence, there is a one-to-one correspondence between the risk and the rate functions [22].

2.1 Non-parametric methods

For discrete observations of time, the survival function is traditionally estimated nonparametrically by the Kaplan-Meier estimator [14]:

$$\hat{S}(t) = \prod_{t_j \le t} \left(\frac{n_j - d_j}{n_j} \right)$$

where n_j is number of individuals in the risk set and d_j is number of events at time t_j . As stated in 1.1.2, an important assumption is that censoring is non-informative. Groups are often compared with different modifications of the log-rank statistic [23, 24]. The cumulative hazard function is typically estimated non-parametrically by the Nelson-Aalen [25, 26] estimator:

$$\widehat{\Lambda}(t) = \sum_{t_j \le t} \frac{d_j}{n_j}$$

In a plot of the Nelson-Aalen estimator versus time, the slope of the plot is an estimate of the hazard rate, e.g. a constant slope indicates a constant hazard. The results from non-parametric methods are presented in graphs and/or tables. Since the estimated distribution is discrete, the continuous hazard function needs to be estimated with the aid of smoothing techniques [27].

2.2 The Cox proportional hazards model

The standard method for regression analysis used to determine the effect of covariates on the hazard function is the Cox Proportional Hazards (PH) model [15]. In this model,

the hazard at time t for individuals with a vector of covariates X and regression coefficients β is:

$$\lambda(t) = \lambda_0(t) \exp\left(\boldsymbol{\beta}^T \boldsymbol{X}\right),$$

where the superscript "T" indicates the transpose of the column vector. Assuming that the baseline hazard function $\lambda_0(t)$ is the same, the hazard (rate) ratio (HR) comparing subjects with covariate values X_1 relative to those with X_0 is:

$$HR = \exp{\{\boldsymbol{\beta}^{\mathrm{T}}(\mathbf{X}_{1} - \mathbf{X}_{0})\}}.$$

Thus, the covariates are assumed fixed over time and the hazards over time are assumed to be proportional, that is, the HR is constant. Although a functional form for the covariates is specified, no particular distribution of survival times is postulated (indeed the baseline hazard function is not even estimated), and so the model is said to be semiparametric. For single event survival data there is a direct relationship between the hazard function and the survival function, hence the Cox PH model can be used to estimate the effect of covariates on the survival function [28]. A stratified Cox PH model is fitted by allowing the baseline hazard function to be different for each value of the stratification variable. The obtained regression estimates are then the same within each stratum. In this scenario, the effect of the stratification variable on the hazard cannot be assessed. Specific estimates of these effects can be obtained by introducing interactions between strata and covariates. One advantage of the Cox PH model is that it is flexible in estimating the effects on the hazard for different values of a covariate [27], but the assumption of proportionality is not always plausible and is often assumed without justification [18]. Furthermore, the model is sensitive to omission of covariates, which can lead to biased regression coefficients or violation of the proportional hazards assumption [27, 29].

2.2.1 Time-dependent covariates

Time-dependent covariates are covariates whose values change over time. They are often classified into external and internal covariates [30] to aid interpretation of the results. External covariates are considered not to be generated by a subject and do not require a subject to be under observation. Examples might include air-pollution or age of a subject. Internal covariates change with the subject, for example a subject's blood pressure over time. Because the values of internal time-dependent covariates are only observed if the subject survives, it is not appropriate to use these covariates in time to event models to make predictions [31]. In the presence of internal time dependent covariates, the relationship between the survival function and hazard function do not hold [31].

When a time dependent covariate x(t) is introduced in the Cox PH regression model the hazard becomes:

$$\lambda(t) = \lambda_0(t) \exp\{\boldsymbol{\beta}^T \boldsymbol{X}(t)\}.$$

Modelling time-dependent covariates as time-fixed can lead to time-dependent bias. Van Walraven et al. [32] conducted a study of papers using survival analysis from a selection of medical journals and found that time-dependent bias was not only common, but that correction of the bias could have changed the conclusion in over half of the studies examined. Beyersmann [33] showed that treating time-dependent covariates as time-fixed generally leads to a smaller hazard ratio in a Cox PH model. However, using time-dependent covariates in the Cox PH model is more complicated than using time-fixed covariates and one has to pay careful attention to how the functional form of the covariates are specified and how the results are interpreted [34].

2.2.2 Time- varying coefficients

One cause of non-proportionality in the Cox PH model occurs when the effect of covariates on the hazard rate varies with time. If this is not accounted for, the results

may be biased [35]. There are several approaches to accommodate time-varying effects. Cox in his original article [15] suggested adding a time by covariate interaction in the model:

$$\lambda(t) = \lambda_0(t) \exp\{\beta x + \beta^* x f(t)\}$$

where f(t) can be any function of time. However, it may be difficult to choose an appropriate function and the wrong choice may result in incorrect interpretation of the results [36]. Other approaches to handle covariates with time-varying coefficients in the Cox PH model include stratifying on the covariate, using a weighted average of the hazard ratio, calculating piecewise constant hazard ratios, and methods based on splines or smooth functions [19, 36].

2.3 The additive Aalen model

An alternative to the Cox PH model is Aalen's non-parametric additive hazard regression model [20]. The model is well suited to handle time-dependent covariates and covariates with a time-varying effect. In this model the hazard for the i^{th} individual with vector of covariates $X_i(t) = (x_{i1}(t), x_{i2}(t), \dots, x_{ip}(t))^T$ is:

$$\lambda_i(t|x_i) = \beta_0(t) + \beta_1(t)x_{i1}(t) + ... + \beta_p(t)x_{ip}(t),$$

where $\beta_0(t)$ is the baseline hazard function and $\beta_j(t)$ (j = 1,..,p) are arbitrary regression functions which measure the effect of the respective covariates [37]. This model allows the effect of the covariates to change over time. In their book, Aalen et al. [37] provide several arguments for the use of this model over the standard Cox PH model. Amongst them are that the hazard ratio as calculated in the Cox PH model might not be a good measure of the effect when the risk factor is rare (risk differences may be better suited), time-dependent covariates are easily incorporated, the model is not vulnerable to neglected covariates, and it may be well suited for analysing excess hazard because the model is not forced to give positive estimates as are proportional hazards models. Despite the obvious advantages, Aalen's additive model is not often used. One of the reasons may be that the regression coefficients are more difficult to estimate and interpret when compared with regression coefficients from the Cox PH model [38].

2.4 Multiple events

In traditional survival analysis, the outcome of interest is the time to occurrence of a single event. Event history analysis is concerned with the occurrence of multiple events over time for an individual. These data have a sequential structure where a subject may experience repeated events of the same type and/or events of different types over time.

The most commonly applied models in medical research for analysing these types of data are marginal [39, 40], conditional [41, 42], shared frailty [43] and multi-state models. For marginal and conditional models the dependence either within individuals in a group or within an individual for recurrent events is accounted for by a robust variance estimator, whereas for shared frailty models the dependence is accounted for by a random effect variable. In multi-state models, the events are seen as a stochastic process describing movements among a finite number of states. The application of these models to multiple event data are described in monographs by, for example, Hougaard [27] and Therneau and Grambsh [44].

The advantage of using multi-state models on survival data is that they easily incorporate time-dependent covariates as transient states, extend from single to multiple event survival analysis and account for left truncation. Multi-state models have been used considerably in analysing data from bone marrow transplant studies [45], and increasingly in cancer studies. In other fields of medical research they are not commonly used despite a rich literature on the theory and their advantages (see for example [27, 37, 46, 47]). One important reason why multi-state models have not been applied more often to survival data in the past may be the lack of available software to implement these models [48], although as this becomes increasingly available, the

adoption of his approach is likely to increase. The next section provides a background on the multi-state modelling approach.

2.4.1 Multi-state models

Time to event data can be modelled as a multi-state stochastic process where an individual occupies a state and moves between states described by the events, that is, when an event occurs the individual changes state (unless prevented by right censoring). States are considered transient, if movement out of them is allowed, or absorbing, if no movement out is allowed. The structure of a multi-state model is often illustrated in diagrams where boxes represent states and arrows represent possible events [48-51].

Considering survival analysis with a single outcome as a multi-state model, an individual can move from a starting state to the outcome state, for example from receiving a hip arthroplasty to having a revision of the arthroplasty (Figure 2.1). Another simple example in the context of competing risks concerns individuals who can move from a starting state into k different absorbing states (see for instance Figure 2.2 where k = 2). In this model the transient state is insertion of an arthroplasty, and dead and revision are considered absorbing states (revision is an absorbing state because time to revision is the outcome of interest). A more complex multi-state model is illustrated in Figure 2.3 which includes three transient states (insertion of an arthroplasty, revision of this arthroplasty and receiving a second arthroplasty) and one absorbing state (dead).



Figure 2.1: Traditional survival model with one event of interest and hazard rate $\lambda(t)$.



Figure 2.2: Competing risks model with two absorbing states and cause specific hazards $\lambda_1(t)$ and $\lambda_2(t)$.



Figure 2.3: Multi-state model with three transient states, one absorbing state (dead) and transition intensities $\lambda_{gh}(t)$ from state g to state h, where g = 0, 1, 2 and h = 1, 2, 3.

The movements between states are called transitions. The instantaneous rate of moving from state g to state h given that the individual is in state g just before time t is called the transition intensity, $\lambda_{gh}(t)$. This is equivalent to the hazard rate in traditional survival analysis with a single event. The transition probability, $P_{gh}(s,t)$, is the probability of moving from state g to h. The state occupation probability, $P_h(t)$ is the probability that an individual is in state h at time t. When all individuals start in state 0 at t = 0, the state occupation probability is the same as the transition probability, $P_{0h}(0,t)$ [37, 49]. If the process is Markovian, the transition probabilities can be calculated directly from the transition intensities, for example by using the Aalen-Johansen estimator [52]. A Markov chain is a process that has the Markov property: the future state depends only on the present state and not any past states. Hence, a Markov model is assumed to be independent of time spent in the current state, if not it is called a semi-Markov model. Further, a Markov model is considered time homogenous if the hazards do not depend on the time (because it is constant), and if it does it is nonhomogenous [48]. In order to investigate if the process is Markovian, a function of current state duration can be included as a time dependent covariate in a Cox PH model (using time since the start of the process as baseline time variable) and testing if the estimated regression coefficient is equal to zero [52]. Datta and Sattten [53] showed that the Aalen-Johansen estimator is valid for estimating state occupation probabilities also in non-Markov models if censoring is independent of state occupied and censoring times. Gunnes et al. [54] showed that the Aalen-Johansen estimator may be valid for less stringent censoring assumptions (i.e. where censoring is allowed to depend on the past) in certain circumstances, such as for small sample sizes.

2.4.2 Competing risks

In survival analysis a competing risks model describes a situation where the occurrence of the event of interest may be precluded by another event, or the risk of the event of interest may be altered by the occurrence of another event [55]. This situation is sometimes (inappropriately) approached by ignoring the competing risks and treating them as censoring events when estimating the survival probability for the event of interest. In this case the assumption of independence of the time to event and the censoring distribution is violated, often leading to a biased Kaplan-Meier estimate and an underestimation of the survival probability. This concept is well illustrated in Putter et al.'s article [51].

There are two mathematical approaches to modelling competing risks: either to model the lifetime as a multivariate random variable or as a multi-state model. The classic competing risks approach is the former: to model the competing risks as parallel latent failure times where there are several potential failure times for each individual. The assumption is that the latent times to failure are independent and that the first failure time determines the whole system failure [56]. The interpretation of the latent failure time is problematic. For example, it is speculative to assume that if a subject does not die from the first cause this will not alter the risk of dying from another cause (which obviously cannot be observed) [27]. This problem was first contemplated in a paper by Bernoulli in 1766 where competing risks were introduced in the context of vaccination for smallpox, see Dietz et al. [57].

The other approach which is increasingly being advocated [49] is to model the competing risks using a multi-state model where subjects pass from an initial state to one of several absorbing states (see for instance Figure 2.2 for k = 2). The transition intensity for moving to one of these states, the instantaneous rate, $\lambda_h(t)$, is called the cause specific hazard rate. The process is Markovian. No assumption is made about independence of failure times but all hazard functions for leaving a state have to be interpreted in the full model [27]. The transition probabilities are known as the cumulative incidences [58]. The cumulative incidence function (CIF) for event type h is:

$$F_h(t) = Pr(T \le t, \varepsilon = h) = \int_{s=0}^{s=t} \lambda_h(s)S(s) \, ds,$$

which can be estimated by the Aalen-Johansen estimator [22, 59]:
$$\hat{F}_{h}(t) = \sum_{t_{j} \le t} \frac{d_{hj}}{n_{j}} \hat{S}(t_{j-1}), \text{ with } t_{1} < t_{2} < \dots t_{r}$$

where $\hat{S}(t_{j-1})$ is the Kaplan-Meier estimator of the survival function considering all events, that is, the probability of being event free prior to t_{hj} , d_{hj}/n_j is the Nelson-Aalen estimator of the cumulative hazard of failure type h where n_j is number at risk at t_j and d_{hj} is the number of events of type h that occur at t_j .

2.4.3 Regression models for competing risks

Estimation of the effect of covariates on the cause specific hazard is often based on a Cox PH model. For example Kalbfleisch and Prentice [30] developed a method where each failure type is modelled separately and the other failure types are censored. Lunn and McNeil [60] developed a model with two versions referred to as method A and method B. In both versions the data are augmented k (failure type) times and a Cox PH model is fitted, method A unstratified and method B stratified. Tai et al. [61] compared four different approaches to analysing competing risks data and developed an extension of the Lunn and McNeil method. However, since the effect of a covariate on the cause specific hazard might be different to the effect on the cumulative incidence function [28, 62], much research has centred on developing regression models for the cumulative incidence function. Klein and Andersen [63] developed a regression method which is based on creating jack-knife pseudo values for the estimate of the cumulative incidence function at specified time points. The obtained pseudo values are then used in generalized estimating equations (GEEs) to model the effect of covariates on the outcome of interest. For the method to work, it is assumed that right censoring is independent on both the multi-state process and of the covariates [64].

The most frequently used method for regression of the cumulative incidence function was developed by Fine and Gray [21]. The model is based on the subdistribution hazard formulated by Gray [65]. For failure from cause 1, i.e. $\varepsilon = 1$, this is:

$$\begin{split} \lambda_1^*(t) &= \lim_{\Delta t \to 0} \frac{1}{\Delta t} P\{t \leq T < t + \Delta t, \varepsilon = 1 | T \geq t \cup (T \leq t \cap \varepsilon \neq 1)\} \\ &= -\frac{\mathrm{d} \log\{1 - F_1(t)\}}{\mathrm{d}t}, \end{split}$$

where $F_1(t)$ is the cumulative incidence function for the failure from cause 1. The risk set associated with the subdistribution hazard consists of individuals who have not experienced the event by time t as well as individuals who have experienced the competing event. The cumulative incidence function and the subdistribution hazard are directly related as in survival analysis with a single event, with:

$$F_1(t)=1-\exp\left(-\int_{s=0}^{s=t}\lambda_1^*(s)ds\right).$$

The Fine and Gray is similar to the Cox PH model in that it is a proportional hazards model of the subdistribution hazard where the effect of the covariates is linear on a complementary log-log transformed cumulative incidence function, i.e.

$$\lambda^*(t) = \lambda_0^*(t) \exp(\boldsymbol{\beta}^T \boldsymbol{X}),$$

where $\lambda^*(t)$ is the hazard of the subdistribution and $\lambda_0^*(t)$ denotes the subdistribution baseline hazard function. As in the Cox PH model the subdistribution baseline hazard function is left unspecified and the subdistribution hazard ratio is:

$$HR_{sub} = \exp\{\beta^{\mathrm{T}}(\mathbf{X}_1 - \mathbf{X}_0)\},\$$

comparing subjects with covariate values X_1 relative to those with X_0 . The model can be used to make predictions of the cumulative incidence function adjusted for covariates. However, because of the definition of the risk set, the subdistribution hazard ratios do not have a direct interpretation unlike the hazard ratios from the Cox PH model [28]. This fact is sometimes not appreciated by researchers, see e.g. [66-68]. The Fine and Gray model has become very popular, partly because of its similarity to the Cox PH model. Further, because commands for implementing the model are now available in common software programs such as Stata and in R, its use will likely continue to increase. The model and its application to arthroplasty registry data are discussed further in Chapter 5.

An overview of the above and other competing risks regression models is provided by Zhang et al. [69] and more recently by Haller et al.[70].

2.5 Analysis of joint replacement registry data

A number of joint replacement registries have been established worldwide. Results from the analysis of registry data are typically presented in annual reports and publications in journals [71, 72].

Data in joint replacement registries are traditionally analysed with survival methods where the outcome of the analysis is 'time to revision', that is, the time interval between the operation date of insertion of the prosthesis and the operation date of revision. Revision is the main indicator of failure of a joint replacement. It is a crude measure of failure as it depends on many factors such as the health of the patient, waiting lists, the surgeon threshold for operation etc., but in its favour it is an unambiguous outcome [73]. The survival function is commonly estimated with the Kaplan-Meier method and results are presented in figures and tables [13, 74]. Patients are censored either administratively (at the end of each year) or because of death. Groups are compared with log-rank tests. Revision rates of different prostheses are compared using the Cox PH model, adjusting for covariates such as age and sex [75].

This approach to analysis is followed for example by the Nordic countries' arthroplasty registries and the Australian registry. The National Joint Registry (NJR) in the United Kingdom use a flexible parametric model [76] to model the hazard rates of revision, thereby obtaining both absolute and relative risks. An extension of the model is also used to estimate the cumulative incidence function and the effect of covariates in the presence of the competing risk of death [77].

Several registries compare risk of revision between different types of prostheses using revision rate per 100 or 1000 component years [78]. This method adjusts for the different lengths of time the prostheses have been at risk of failure, but it does not provide information on how the risk of revision may change over time.

When data in the AOA NJRR are analysed, proportionality in the Cox PH model is checked and, if evidence of a time varying effect is found, time points are chosen iteratively for calculating hazard ratios for each selected time period.

As the main outcome of interest in analysis of joint replacement data is failure of the prosthesis, data are analysed at the prosthesis-level. The consequence is that a patient who has several joint replacements may appear several times in the same analysis. For the purpose of data management and linkage, the NJR have restructured their data from procedure-level to person-level, but the data are analysed at procedure-level. With increasing amounts of data in joint replacement registries, the system for data management will become increasingly important.

This Chapter has provided an introduction to time to event analysis. The relevant literature has been reviewed and some issues related to the analysis of event histories have been discussed. In Chapters 4 to 7, these issues are considered in more detail, specifically with respect to the analysis of joint replacement registry data.

3 DATA SOURCES

Arthroplasty is surgical reconstruction or replacement of a joint. In this thesis, arthroplasty means joint replacement surgery. In addition to collecting data on hip and knee replacements, some arthroplasty registries also collect data on replacements of other joints such as elbow, shoulder, ankle and fingers. This thesis considers only data on hip and knee replacements, but the statistical methods discussed are applicable to data on other types of joint replacements.

There are two main categories of hip replacement, total hip replacement and partial hip replacement. In the former, the articular surfaces of both the acetabulum and femur are replaced. The main indication for this procedure is symptomatic osteoarthritis. In partial hip replacement (hemiarthroplasty) only the femoral component is replaced. This procedure is mainly performed as treatment for fractured neck of femur.

The main categories of knee replacements are partial and total knee replacements. In partial knee replacements only parts of the femur and/or tibia articular surface are replaced whereas in total knee replacement the entire femorotibial articulation is replaced with a single femoral and a single tibial prosthesis. Again, the main indication is symptomatic osteoarthritis.

In the following I provide a short description of the two registries used as data sources for this thesis, the Australian Orthopaedic Association National Joint Replacement Registry and the Norwegian Arthroplasty Register.

3.1 The Australian Orthopaedic Association National Joint Replacement Registry

The Australian Orthopaedic Association National Joint Replacement Registry (AOA NJRR) was established and began data collection in 1999, but was not fully national until 2003. The need to establish a joint replacement registry in Australia was inspired

by the success of the Swedish Knee Arthroplasty Registry in improving standards and reducing costs of joint replacement surgery. The purpose of the AOA NJRR is to "define, improve and maintain the quality of care for individuals receiving joint replacement surgery"[13].

The AOA NJRR is owned by the Australian Orthopaedic Association and its technical support is provided by the Data Management & Analysis Centre (DMAC), University of Adelaide. Until 2008, the AOA NJRR was only collecting data on hip and knee replacements, but since then data on other joints (shoulder, elbow, ankle, etc.) replacements have been included. Data are recorded on both primary procedures and revisions. The Registry now (as of December 2012) has data on more than 700,000 procedures. Almost 100% of hip and knee procedures performed in Australia are reported. The amount of data recorded for each procedure is relatively small, which contributes to good compliance and reporting. The AOA NJRR obtain mortality data on patients who have received joint replacements (which are necessary for the appropriate analysis of the data) from the National Death Index, a data base maintained by the Australian Institute of Health and Welfare.

Information from the analysis of the Registry data is disseminated through annual and supplementary reports. There is an online facility where surgeons can access their individual data and for prosthesis companies and regulatory bodies to monitor prostheses.

Up until 31 December 2011, 332,351 hip procedures in 273,534 individual patients were recorded in the AOA NJRR. Approximately 12% of these procedures were revisions and 16% of the primary hip procedures were partial. Of 36,503 bilateral hip procedures, 4.6 % of patients had both hips operated upon the same day and 20% between 1 day and 6 months. The number of patients who received a hip replacement and have since died is 57,586 (21.1%).

For knee procedures, 380,726 were reported in 285,474 individual patients, 8% of which were revisions. The number of bilateral knee procedures was 72,788, 22.5% of

patients had both knees operated upon the same day and 13% performed between 1 day and 6 months. The number of patients who received a knee replacement and have since died is 27,597 (9.7%)[13].

3.2 The Norwegian Arthroplasty Register

The Norwegian Hip Arthroplasty Register (NAR) was established in 1987 in response to high failure rates of some prostheses that had been introduced to the market without prior clinical studies [79]. In 1994 it was extended to register all types of joint replacements performed in Norway. The main purpose of the NAR is to "function as a surveillance tool to identify inferior implants as early as possible" [80].

The NAR has excellent coverage. It has been reported that the NAR registered 97% of hip arthroplasties patient recorded in the Norwegian Patient Register 1999-2002 [81, 82]. The data are linked through patients' national social security number to the Norwegian Population Register so that information about emigration and death are obtained when the NAR data base is updated.

The 2012 annual report [74] includes 162,971 hip procedures performed between 1987 and 2012; approximately 14% of these were revision procedures. In 2011, 7,360 primary hip procedures were performed. In the period 1994 to 2012, 51,337 knee procedures were recorded, 8% of these were revision procedures. In 2011, 4,526 primary knee procedures were performed.

3.3 Ethical considerations

The AOA NJRR is a Declared Federal Quality Assurance Activity and is required by law to abide by certain specified requirements, for example, protect information that identifies individuals in the AOA NJRR. It has been subjected to ethics approval. The AOA NJRR only releases de-identified data. The data released for this project had the patients' identifiers removed. Further, there were no other characteristics in the data used in this thesis from which to identify any particular patient or surgeon. Prior to undergoing their arthroplasty patients are informed about the AOA NJRR and informed that they may 'opt-off', that is, choose not to be included in the Registry database. They are given a phone number to ring for further information. Surgeons are not identified on the database unless they choose to be.

The data obtained from the NAR are also de-identified. The NAR obtains permission from each patient to collect information on their procedure.

4 COMPETING RISKS SURVIVAL ANALYSIS APPLIED TO DATA FROM THE AUSTRALIAN ORTHOPAEDIC ASSOCIATION NATIONAL JOINT REPLACEMENT REGISTRY

4.1 Preface

This chapter contains the first of four articles that contribute to this thesis. The article has been published in *Acta Orthopaedica* [83]. It investigates the use of the Kaplan Meier (KM) method and the Cumulative Incidence Function (CIF) in the analysis of arthroplasty registry data. The aim of the study was to compare the estimates of risk of revision from the two methods while treating death as a competing risk. The purpose was to demonstrate the degree to which the risk of revision is biased using the KM method, compared to estimates using competing risks methods, on subsets of data from the AOA NJRR with different incidences of the competing risk death.

The article also explains the problems associated with using the KM method in competing risks scenarios and reasons why KM estimates are often biased.

Of note is that the competing risks model is equivalent to a simple multi-state model with two or more absorbing states and one transient state. Multi-state models with a more complex structure are described in Chapters 6 and 7.

4.2 Statement of Authorship

Gillam MH, Ryan P, Graves SE, Miller LN, de Steiger RN, Salter A. Competing risks survival analysis applied to data from the Australian Orthopaedic Association National Joint Replacement Registry. *Acta Orthop* 2010;81(5):548-55. Doi: 10.3109/17453674.2010.524594

Marianne H Gillam (Candidate)

Designed the study, performed all analysis, interpreted the results, drafted the manuscript and acted as corresponding author.

Signed: Date:

.....

27th December 2012

Philip Ryan

Contributed to the design of the study, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Doto:	
 Date.	************************
	Date:

Stephen E Graves

Contributed to the acquisition of the data, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed: Date:

Lisa N Miller

Contributed to the acquisition of the data, interpretation of the results, and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:	690		Date:	28/12/12
---------	-----	--	-------	----------

29

Richard N de Steiger

.....

Contributed to the acquisition of the data, interpretation of the results, and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:

Date:

8.1.13

Amy Salter

Contributed to the interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed: Date:

4.3 Article

Marianne H Gillam 1, Philip Ryan, Stephen E Graves, Lisa N Miller, Richard N de Steiger, and Amy Salter (2010) Competing risks survival analysis applied to data from the Australian Orthopaedic Association National Joint Replacement Registry. *Acta Orthopaedica, v. 81 (5), pp. 548-555, October 2010*

NOTE: This publication is included in the print copy of the thesis held in the University of Adelaide Library.

It is also available online to authorised users at:

http://dx.doi.org/10.3109/17453674.2010.524594

4.4 Additional Discussion

The data in the published article were obtained from the AOA NJRR. The Australian registry achieved full national coverage in 2002 and at the time of analysis the data consisted of records covering a 7-year period. In the discussion in the article, it was pointed out that scope for further research would entail applying competing risks methods to data with a much longer follow-up time.

The same analysis was subsequently performed on data from the Norwegian Arthroplasty Register (NAR) which has records on hip replacements in Norway from 1987. The data consist of 78,102 first recorded hip arthroplasties in patients who received THA for OA from 1987 to 2010. As in the published article, results were compared for patients younger than 70 years with those from patients who were 70 years or older. These results are presented in Table 4.5 and Figures 4.8 and 4.9 below. The KM and CIF estimates, differences and relative differences are similar to the estimates based on the Australian data at 1 year and at 5 years (Table 4.4). Because mortality in this group of patients was relatively low for the first 5 years after receiving the arthroplasty, there was little difference between the KM and CIF estimates of risk of revision at these time points. However, with increasing mortality over time with increasing age of the patients, it is evident that the KM method eventually overestimates the risk of revisions to a substantial degree in this group, as was the case in the group of patients from the AOA NJRR who received arthroplasty for fractured neck of femur. For example, after 20 years the relative difference in the youngest age group was 21.5% whereas in the oldest age group it was 68.9% (Table 4.5). Because CIF estimates of the risk of the event of interest should be interpreted in conjunction with estimates of the competing event(s), the figures for the estimates of risk of death over the period are also presented (Figure 4.9). The CIF estimates are somewhat lower than the KM estimates for the risk of death and represents estimates of the risk of death over time without first having had a revision. When revision is not considered a competing event to death, which is most commonly the case, the KM estimates are the appropriate estimates of risk of death at specific points in time. The results demonstrate that even in a population with relative low mortality such as individuals who have received THA for OA [101], accounting for competing risks is important in analysis with long follow-up time.

 Table 4.5: Data from the Norwegian Arthroplasty Register. Percent estimates (with

 95% confidence interval) of revision in patients with osteoarthritis who underwent

 total hip replacement – by age group.

	1 Yr	5 Yrs	10 Yrs	15 Yrs	20 Yrs		
Age < 70 years							
At risk	31,634	22,010	12,211	5,413	1,549		
KM ^a	0.8 (0.7-0.9)	4.2 (4.0-4.5)	10.8 (10.4-11.3)	19.0 (18.3-19.7)	25.3 (24.3-26.3)		
CIF ^b	0.8 (0.7-0.9)	4.2 (3.9-4.4)	10.2 (9.8-10.6)	16.8 (16.2-17.3)	20.8 (20.1-21.5)		
Diff ^c	0.002	0.08	0.62	2.21	4.47		
RD ^d	0.2%	1.9%	6.2%	13.2%	21.5%		
Age ≥70 years							
At risk	40,398	27,552	12,243	3,573	514		
KM	1.0 (0.9-1.1)	3.4 (3.2-3.6)	6.3 (6.0-6.5)	9.2 (8.8-9.8)	12.1 (11.1-13.2)		
CIF	1.0 (0.9-1.0)	3.2 (3.1-3.4)	5.3 (5.1-5.5)	6.6 (6.3-6.9)	7.2 (6.9-7.5)		
diff	0.01	0.2	0.94	2.64	4.95		
RD	0.9%	5.7%	17.7%	39.9%	68.9%		

^a Kaplan-Meier estimate of Cumulative Percent Revised.

^b Cumulative Incidence Function.

^c Difference (bias of the KM estimate).

^d Relative Difference (bias of KM estimate relative to the CIF).







Figure 4.9: Estimates of death by age group in patients with OA and THA (data from the NAR)

5 DIFFERENT COMPETING RISKS MODELS APPLIED TO DATA FROM THE AUSTRALIAN ORTHOPAEDIC ASSOCIATION NATIONAL JOINT REPLACEMENT REGISTRY

5.1 Preface

In the previous chapter, non-parametric methods for estimating the risk of revision in the presence of the competing risk of death were compared. It was shown that when the risk of death was high, the KM method substantially overestimated the risk of revision compared to estimates using competing risks methods, and that the bias increased with time as the incidence of the competing risk of death increased.

This chapter deals with estimating the effect of covariates on the hazard rate and on the risk of revision and death. The hazard rate of revision is the instantaneous risk of revision at any given time whereas the risk of revision, as estimated by the CIF, is the accumulated risk before a certain time. Since the latter estimates depend on the hazard rates of both death and revision, the effects of covariates on the rate and risk of revision may differ.

In the following article, which has been published in *Acta Orthopaedica* [102], the aim was to examine the use of different models in analysing arthroplasty registry data in the presence of competing risks. It illustrates the use of methods for assessing the effect of covariates on the revision rate and methods for assessing the effect on the risk of revision. Methods for handling covariates with time dependent effects are also addressed, and guidance is provided on how to interpret estimates from competing risks analysis.

5.2 Statement of authorship

Gillam MH, Salter A, Ryan P, Graves SE. Different competing risks models applied to data from the Australian Orthopaedic Association National Joint Replacement Registry. *Acta Orthop* 2011;82(5):513-20. Doi: 10.3109/17453674.2011.618918.

Marianne H Gillam (Candidate)

Designed t he s tudy, pe rformed all a nalysis, i nterpreted t he r esults, dr afted t he manuscript, and acted as corresponding author

Signed: Date:

Amy Salter

Contributed to the design of the study, interpretation of the results, and reviewed the manuscript. I g ive consent f or Marianne Gillam to present this paper t owards examination for the Doctor of Philosophy.

Signed: Date:

Philip Ryan

Contributed to the design of the study, interpretation of the results, and reviewed the manuscript. I g ive consent f or Marianne Gillam to present this p aper towards examination for the Doctor of Philosophy.

Signed: Date:

Stephen E Graves

Contributed to the acquisition of the data, interpretation of the results, and reviewed the manuscript. I gi ve c onsent f or M arianne G illam to p resent this paper to wards examination for the Doctor of Philosophy.

Signed: Date: 27th December 2012.....

5.3 Article

Marianne H Gillam, Amy Salter, Philip Ryan, and Stephen E Graves (2011) Different competing risks models applied to data from the Australian Orthopaedic Association National Joint Replacement Registry

Acta Orthopaedica, v. 82 (5), pp. 513-520, October 2011

NOTE: This publication is included in the print copy of the thesis held in the University of Adelaide Library.

It is also available online to authorised users at:

http://dx.doi.org/10.3109/17453674.2011.618918

5.4 Additional discussion

In tra ditional s urvival analysis with a s ingle out come there i s a one -to-one correspondence between the hazard rate and the survival function so that the effect of covariates as estimated by the Cox PH model corresponds directly to the effect on the survival f unction [22]. T he previous a rticle demonstrates t hat in c ompeting r isks scenarios the effect of covariates on the risk of revision may be different to the effect of the hazard rate of revision. This is evident for example when comparing the effect of sex o n revision: m ales h ad a si gnificantly h igher h azard o f r evision t han f emales whereas there was no evidence of a d ifference between the sex es in the actual risk of revision. In a competing risks scenario with one event of interest and one c ompeting event and with cau se specific h azards $\lambda_1(t)$ and $\lambda_2(t)$ respectively, the r elationship between $\lambda_1(t)$ and the subdistribution ha zard $\lambda_1^*(t)$ for the event of interest is [121, 122]:

$$\lambda_1(t) = \left(1 + \frac{P(T \le t, \varepsilon = 2)}{P(T > t)}\right) \times \lambda_1^*(t) = \left(1 + \frac{F_2(t)}{P(T > t)}\right) \times \lambda_1^*(t),$$

where $F_2(t)$ is the CIF of the competing event and

$$P(T > t) = \exp\left(-\int_{s=0}^{s=t} \lambda_1(s) + \lambda_2(s)ds\right)$$

Proportionality of either the Cox PH model or the Fine and Gray model does not imply proportionality of the other. However, even if the model for the subdistribution hazard is misspecified, it may still be useful for estimating average effects of covariates on the CIF [123, 124].

The relationship between the hazard ratios (HRs) from the Cox PH model and the Fine and Gray model are described by Lau et al. [109]. Table 5.3 shows how the HR of the event of interest relates to the subdistribution HR in a scenario with one competing risk. For example, if the HRs indicate a decreasing effect of a covariate (HR < 1) on the cause specific hazards for both the event of interest and the competing event, the subdistribution HR will be larger than the cause specific HR for this event (Table 5.3). In general, it appears that if a covariate influences both cause specific hazards in the same direction, it can potentially result in the covariate having an opposite effect on the hazard rate to the actual risk of an event. See also articles by Allignol et al. and Dignam et al. [125, 126].

The relationship between the cause specific hazards and the subdistribution hazard is further complicated in that the numerical values of the subdistribution HRs cannot be interpreted quantitatively, unlike HRs obtained from Cox PH models, but only as qualitative relative effects on the CIF [22]. This is because the risk set associated with the subdistribution hazard for the event of interest at any time consists of individuals who have not experienced any event in addition to individuals who have experienced the competing event(s) [21].

Table 5.4 provides results from a stratified Cox PH model and the modified Fine and Gray model applied to the registry data from the AOA NJRR in the previous article (Table 5.4 is similar to Table 5.2 in the article except that it also includes HRs for death). The results in table 5.4 are consistent with the relationship between the cause specific HRs and the subHRs as described in Table 5.3 below.

HR ₁	HR ₂	
< 1	< 1	$subHR_1 > HR_1$
< 1	>1	$subHR_1 < HR_1$
>1	< 1	$subHR_1 > HR_1$
>1	>1	$subHR_1 < HR_1$

Table 5.3: Relationship between HRs and subHRs, modified from Lau et al. [109]

 HR_1 and HR_2 : cause specific hazard ratios for event of interest and competing event respectively. $subHR_1$: subdistribution hazard ratio for event of interest.

Table 5.4: Cause specific hazard ratios (HRs) and subdistribution hazard ratios (subHRs) for different covariate for a stratified CoxPH model and a modified Fine and Gray model respectively.

Models	Cox PH				Modified Fine and Gray	
Event type	Revision		Death		Revision	
	HR (95% CI)	Р	HR (95% CI)	Р	subHR (95% CI)	Р
Age: young vs. old ^a	1.28 (1.05, 1.56)	0.01	0.81 (0.77, 0.86)	< 0.001	1.36 (1.10, 1.67)	0.004
Male vs. female ^b	1.37 (1.10, 1.70)	0.005	1.94 (1.83, 2.05)	< 0.001	1.94 (0.83, 1.31)	0.73
Fixation type	stratified	-	stratified	-	time dependent	-
Monoblock vs. bipolar ^b	1.45 (1.09, 1.94)	0.01	1.85 (1.71, 1.99)	< 0.001	1.30 (0.97, 1.74)	0.08
Unipolar vs. bipolar ^c	1.38 (1.02, 1.88)	0.04	1.04 (0.96, 1.14)	0.36	1.44 (1.04, 1.98)	0.03
Unipolar vs. monoblock ^d	0.95 (0.74, 1.23)	0.71	0.56 (0.52, 0.61)	< 0.001	1.11 (0.85, 1.45)	0.46
$^{a}HR_{revision} > 1 \text{ and } HR_{death} < 1 \text{ HR}, subHR_{revision} > HR_{revision}$						
$^{b}HR_{revision} > 1 \text{ and } HR_{death} > 1 \text{ HR}, subHR_{revision} < HR_{revision}$						
$^{c}HR_{revision} > 1$ and $HR_{death} \approx 1$ HR, $subHR_{revision} > HR_{revision}$						
$^{d}HR_{revision} < 1$ and HR_{dec}	_{ath} < 1 HR, sub	bHR _{revi}	$_{sion} > HR_{revision}$			

6 MULTI-STATE MODELS AND ARTHROPLASTY HISTORIES AFTER UNILATERAL TOTAL HIP ARTHROPLASTIES

Introducing the Summary Notation for Arthroplasty Histories

6.1 Preface

In the two previous chapters methods for analysing arthroplasty registry data in the presence of competing risks have been applied and discussed. The competing risks model is an example of a simple multi-state model with two or more absorbing states. In more complex multi-state models individuals can move into a finite number of different transient states before reaching absorbing states.

With a well documented increasing proportion of elderly people in the population and an increasing incidence of joint replacements in the population, progressively more individuals are expected to experience several joint procedures during their lifetime. The following article, which has been published in *Acta Orthopaedica* [127] with supplementary article data, investigates the use and suitability of multi-state modelling techniques in analysing data on complex arthroplasty histories. The aim of this article was to develop a model that could be used to describe arthroplasty histories and estimate transition intensities and probabilities associated with multiple joint procedures in an individual.

As joint replacement registries grow larger over time, management of the resultant data becomes increasingly complex. Joint replacement registries record each primary joint replacement with associated revisions, such that each individual may be recorded several times. Another aim of this article was to develop a notation that would enable identification and management of the complete arthroplasty history of each individual.

6.2 Statement of Authorship

Gillam MH, Ryan P, Salter A, Graves SE. Multi-state models and arthroplasty histories after unilateral total hip arthroplasties. *Acta Orthop* 2012;83(3):220-6.

Doi: 10.3109/17453674.2012.684140.

Marianne H Gillam (Candidate)

Designed t he s tudy, pe rformed all a nalysis, i nterpreted t he r esults, dr afted the manuscript, and acted as corresponding author

Signed: Date:

Philip Ryan

Contributed to the design of the study, interpretation of the results, and reviewed the manuscript. I g ive c onsent f or Marianne Gillam to p resent this p aper to wards examination for the Doctor of Philosophy.

Signed: Date:

Amy Salter

Contributed to the design of the study, interpretation of the results, and reviewed the manuscript. I g ive c onsent f or Marianne Gillam to p resent this p aper to wards examination for the Doctor of Philosophy.

Signed: Date:

Stephen E Graves

Contributed to the acquisition of the data, interpretation of the results, and reviewed the manuscript. I g ive c onsent f or Marianne Gillam to p resent this p aper to wards examination for the Doctor of Philosophy.

27th December 2012 Signed: Date:

6.3 Article

Marianne H Gillam, Philip Ryan, Amy Salter and Stephen E Graves (2012) Multi-state models and arthroplasty histories after unilateral total hip arthroplasties: introducing the Summary Notation for Arthroplasty Histories. *Acta Orthopaedica, v. 83 (3), pp. 220-226, June 2012*

NOTE: This publication is included in the print copy of the thesis held in the University of Adelaide Library.

It is also available online to authorised users at:

http://dx.doi.org/10.3109/17453674.2012.684140

7 THE PROGRESSION OF END-STAGE OSTEOARTHRITIS: ANALYSIS OF DATA FROM THE AUSTRALIAN AND NORWEGIAN JOINT REPLACEMENT REGISTRIES USING A MULTI-STATE MODEL

7.1 Preface

In Chapter 6 the use of a multi-state model for analysing the rates and probabilities of revision and receiving a second arthroplasty was investigated. The Summary Notation for Arthroplasty Histories, SNAH, was developed as a tool to manage data in arthroplasty registries on patients with multiple joint procedures and also to facilitate multi-state modelling. This chapter contains an article which has been accepted (16th December 2012) for publication in *Osteoarthritis and Cartilage* [135] where rather than focussing on revision of prostheses, a multi-state modelling technique is applied to the investigation of progression of osteoarthritis. Osteoarthritis is a common chronic disease and the pathogenesis is not clear. Symptomatic osteoarthritis is by far the most common indication for joint replacement surgery. The article investigates the progression of osteoarthritis. The aim of the article was to determine if evidence of a pattern in the progression of osteoarthritis in large weight bearing joints could be found in independent data from two large national joint replacement registries.

7.2 Statement of Authorship

Gillam MH, Lie SA, Salter A, Furnes O, Graves SE, Havelin LI, Ryan P. The progression of end-stage osteoarthritis: Analysis of data from the Australian and Norwegian joint replacement registries using a multi-state model.

Accepted for publication in Osteoarthritis and Cartilage 17 December 2012

Marianne H Gillam (Candidate)

Designed the study, performed all analysis, interpreted the results, drafted the manuscript and acted as corresponding author.

Signed:

Date:

Stein Atle Lie

Contributed to the acquisition of the data, design of the study, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:

Date:

Ver 20th 2012

Amy Salter

Contributed to the design of the study, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:

Date:

....

Ove Furnes

Contributed to the acquisition of the data, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy

Signed:

Date:

2nd January 2013

Stephen E Graves

Contributed to the acquisition of the data, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:

Leif I Havelin

Contributed to the acquisition of the data, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:

Date:

Date:

December 20. 2012

December 27th 2012.

Philip Ryan

Contributed to the design of the study, interpretation of the results and reviewed the manuscript. I give consent for Marianne Gillam to present this paper towards examination for the Doctor of Philosophy.

Signed:

Date:

7.3 Article

7.3.1 Abstract

Objective: The incidence of joint replacements is considered an indicator of symptomatic end-stage osteoarthritis (OA). We analysed data from two national joint replacement registries in order to investigate whether evidence of a pattern of progression of end-stage hip and knee OA could be found in data from large unselected populations.

Design: We obtained data on 78,634 hip and 122,096 knee arthroplasties from the Australian Orthopaedic Association National Joint Replacement Registry and 19,786 hip and 12,082 knee arthroplasties from the Norwegian Arthroplasty Register. A multistate model was developed where individuals were followed from their first recorded hip or knee arthroplasty for OA to receiving subsequent hip and/or knee arthroplasties. We used this model to estimate relative hazard rates and probabilities for each registry separately.

Results: The hazard rates of receiving subsequent arthroplasties in non-cognate joints were higher on the contralateral side than on the ipsilateral side to the index arthroplasty, especially if the index was a hip arthroplasty. After 5 years, the estimated probabilities of having received a knee contralateral to the index hip were more than 1.7 times the probabilities of having received a knee ipsilateral to the index hip.

Conclusion: The results indicate that there is an association between the side of the first hip arthroplasty and side of subsequent knee arthroplasties. Further studies are needed to investigate whether increased risk of receiving an arthroplasty in the contralateral knee is related to having a hip arthroplasty and/or preoperative factors such as pain and altered gait associated with hip OA.

7.3.2 Introduction

Osteoarthritis (OA) is a common chronic disease, leading to chronic pain, decreased quality of life and disability [136]. OA often involves multiple joints and the greatest disability is caused by hip and knee OA [137] for which joint replacement is often a successful treatment. The pathogenesis of OA is not clear. It is thought to be a combination of genetic factors, systemic risk factors and biomechanical factors [138-140]. The sequence of progression of OA to different joints can inform the understanding of the pathogenesis of OA. The incidence of joint replacements is considered by many an indicator of symptomatic end-stage OA [129, 141, 142], hence the progression of joint replacements in individuals is an indicator of the progression of end-stage OA. Evidence suggests that the pattern of progression of end-stage OA in large weight bearing joints is not a random process. For example, Shakoor et al. [129] found that a greater proportion of individuals who had received total hip arthroplasty (THA) or total knee arthroplasty (TKA) for OA, received their second arthroplasty in the cognate contralateral joint. Of those individuals who had received a unilateral THA followed by a TKA, a higher proportion received an arthroplasty in the contralateral knee than in the ipsilateral knee. This was in contrast to individuals with rheumatoid arthritis, where there was no difference between the sides of TKA following a THA.

Using joint replacements as an indicator of symptomatic end-stage OA, data from population-based arthroplasty registries can provide information on the progression of end-stage OA in large weight bearing joints. The objective of this study was to investigate whether evidence of a pattern of progression of joint replacements in large weight bearing joints could be found in independent data from two large national joint replacement registries using a multi-state model for each registry separately. The study hypothesis was that there is an association between the side of the first hip or knee arthroplasty and the side of subsequent arthroplasties in non-cognate large weight bearing joints.

7.3.3 Material and methods

We obtained data from the Australian Orthopaedic Association National Joint Replacement Registry (AOA NJRR) and the Norwegian Arthroplasty Register (NAR). The Norwegian Arthroplasty Register and the Australian Orthopaedic Association National Joint Replacement Registry are national registries that record and analyse data on subjects who have received joint replacements. The NAR has collected data on hip arthroplasties since 1987 and knee arthroplasties since 1994 [79]. The AOA NJRR started collecting data on hip and knee arthroplasties in 1999 and became national in 2002 [143]. The NAR captures 97% of all hip and knee replacements performed in Norway [81]. The AOA NJRR also has excellent coverage, after validation of its records against state hospital data, the AOA NJRR obtains an " almost complete dataset relating to hip and knee replacement in Australia"[143].

We obtained data on subjects who had received a first recorded hip or knee arthroplasty for OA in the period from January 1, 2002 to December 31, 2010 from the AOA NJRR and the NAR. Individuals who had received a hip or a knee arthroplasty before January 1, 2002 were excluded, as were individuals who were registered with a revision but without a primary arthroplasty. We also excluded individuals who had received two arthroplasties on the same day because the focus of the study was progression of OA. Some patients could have received arthroplasties prior to the time that the NAR and the AOA NJRR were established. Including these patients in the study sample would lead to inflation of the risk set and potentially bias the estimates. In order to minimise this complication, especially with regard to the Australian data, patients aged 55-74 years were selected because individuals within this age group compared to older individuals were less likely to have received an arthroplasty prior to 2002 that was not recorded in the joint registries. The lower age limit was selected because younger individuals had low prevalence of OA compared to the selected age group. For descriptive purposes, individuals were categorised into two groups based on age (55-64 years and 65-74 years).

The arthroplasty history of interest consisted of four possible arthroplasties; two hips and two knees. We developed a multi-state model where patients were followed as they moved through different possible states from a first arthroplasty (either hip or knee) to receiving subsequent hip or knee arthroplasties, death or until study closure (right censored). The states describe conditions such as having had a joint replacement. When an event occurs, such as receiving a joint replacement, the individual changes state. Once the structure of the multi-state model is specified it can provide probabilities and hazard ratios (HRs) associated with states and with movements from one state to another [52].

The model with 14 possible states that can be occupied (boxes), and paths (arrows) that can be travelled, is illustrated in Figure 7.1. The-starting point for an individual is any one of four possibilities (left hip, right hip, left knee, right knee). After the first arthroplasty there is a total of three possible subsequent primary arthroplasties for an individual (contralateral cognate, left non-cognate, right cognate). The possibilities for the second primary arthroplasty is therefore one of the three remaining hip(s)/knee(s). The possibilities for the third primary arthroplasty is one of the remaining two hips(s)/knee(s) and so forth. At any time subjects could enter a so-called "absorbing" state, being dead (we adopt the naming convention that 'death' is an event and being 'dead' is a state [50]). Because the aim of the study was to investigate the progression of joint replacements for OA, individuals who received subsequent arthroplasties for other indications (e.g. fractured neck of femur) were merged with the state dead. The use of multi-state models and notation in analysing complex arthroplasty histories are described in more detail in a previous paper [127].



Figure 7.1: Multi-state model

A Cox proportional hazards model [15] was used to estimate the effect of covariates on the transition hazards between states in the model, that is, the instantaneous risk (rate) of a subject moving from one state to another at a given point in time, conditional on being at risk for that particular transition at the time. In order to choose time scale in the model, preliminary analyses were performed to assess if the processes were Markov, that is, if the hazard rates were independent of past states and time spent in current state [144]. Time spent in previous states and in current states were included as covariates in the model and the results indicated that time spent in the current state, but not in the previous state affected the transition hazards. Therefore a model was chosen where time was reset (clock-reset model or semi-Markov [51]) after entering a new state. The Cox model was stratified on transitions such that transition hazards were calculated for each possible transition and the covariates were transition specific. The covariate of primary focus was the side (right or left) of the first arthroplasty as we wished to assess if the hazards of subsequent transitions were dependent on whether subjects received their first arthroplasty on the right side or on the left side. The HRs were adjusted for age, sex and which joint had a revision (time dependent covariate). The proportional hazards assumption in the Cox model was checked with Schoenfeld residuals for each transition and found to be satisfactory.

To illustrate the possible states through which individuals could move, the full model is presented in Figure 7.1. However, we only present HRs that are relevant for the aim of the study, that is, HRs that compare the effect of side of the first arthroplasty (right vs. left) on transitions to arthroplasties in non-cognate joints. The transition paths and states of interest are highlighted Figure 7.1.

In order to further assess if there was a difference in the absolute risk of having received an arthroplasty in a hip or knee followed by an arthroplasty in a non-cognate joint at different points in time, we estimated the state probabilities for transitions from state 1 to state 3 and from state 1 to state 4 using the Aalen-Johansen estimator [59].

Observations were right-censored on December 31, 2010 after the last event (after last arthroplasty) if death had not yet occurred. For the data preparation and analyses we used the 'mstate' package [128] in the software environment 'R' [108] and Stata version 11.

7.3.4 Results

There were 200,730 subjects included from the AOA NJRR and 31,868 subjects from the NAR. The distribution of patients by type of first arthroplasty (hip or knee), country, age, sex and side of first arthroplasty is presented in Table 7.1. The data contain records on 98,420 first hip arthroplasties and 134,178 first knee arthroplasties. In Australia more subjects received first knee than first hip arthroplasties (61% vs. 39%), whereas in Norway more subjects received first hip than first knee arthroplasties (62% vs. 38%). For both countries there were more first arthroplasties on right sides with this being most pronounced for first hip arthroplasties from Norway. The Norwegian data had a lower proportion of males than females, especially for hip arthroplasties. In the Australian data, this was also the case for knee arthroplasties, whereas in the hip data there were equal proportions of males and females. For both hip and knee arthroplasties there were more subjects in the oldest age group.

		First arthroplasty hip		First arthro	lasty knee	
		Australia	Norway	Australia	Norway	
Age:	55-64 years	34,093 (43%)	7,691 (39%)	50,854 (42%)	5,286 (44%)	
	65-74 years	44,541 (57%)	12,095 (61%)	71,242 (58%)	6,796 (56%)	
Sex:	Males	39,435 (50%)	6,816 (34%)	54,462 (45%)	4,554 (38%)	
	Females	39,199 (50%)	12,970 (66%)	67,634 (55%)	7,528 (62%)	
Side:	Left	34,715 (44%)	8,003 (40%)	55,149 (45%)	5,447 (45%)	
	Right	43,919 (56%)	11,783 (60%)	66,947 (55%)	6,635 (55%)	
Total		78,634	19,786	122,096	12,082	

Table 7.1: Distribution of individuals according to covariates.

	First arthroplasty hip		First arthroplasty knee		
-	Australia	Norway	Australia	Norway	
	n (%*)	n (%*)	n (%*)	n (%*)	
1 arthroplasty	78,634	19,786	122,096	12,082	
	(100)	(100)	(100)	(100)	
No event	58,303 (74)	14,414 (73)	87,874 (72)	8714 (72)	
state $1 \rightarrow$ state 2	12,668 (16)	3867 (20)	26,433 (22)	2521 (21)	
state $1 \rightarrow$ state 3	1828 (2)	228 (1)	1348 (1)	133 (1)	
state $1 \rightarrow$ state 4	2072 (3)	257 (1)	1772 (1)	222 (2)	
state $1 \rightarrow$ state 5	3763 (5)	1020 (5)	4669 (4)	492 (4)	
state $2 \rightarrow$ state 6	172 (1)	26 (1)	268 (1)	27 (1)	
state $2 \rightarrow$ state 7	232 (2)	46 (1)	335 (1)	46 (2)	
state $3 \rightarrow$ state 6	127 (7)	14 (6)	134 (10)	10 (8)	
state $3 \rightarrow$ state 8	208 (11)	26 (11)	159 (12)	23 (17)	
state $4 \rightarrow$ state 7	153 (7)	26 (10)	196 (11)	17 (8)	
state $4 \rightarrow$ state 8	241 (12)	33 (13)	158 (9)	27 (12)	
state 2-4 → state 9- 11	574 (3)	154 (4)	952 (3)	94 (3)	
state $6-8 \rightarrow state 12$	97 (10)	20 (12)	127 (10)	8 (5)	
state $6-8 \rightarrow state 13$	20 (2)	3 (2)	25 (2)	3 (2)	
state $12 \rightarrow$ state 14	1 (1)	0 (0)	4 (3)	0 (0)	

Table 7.2: Numbers and percent of events in the multi-state model (Figure 7.1) at the end of the study period for patients whose first arthroplasty was a either a hip or a knee arthroplasty for OA.

*Percent of number of individuals who entered the state

Table 7.2 shows the numbers and proportions of arthroplasty events that had occurred at the end of the study period. Between 72% and 74% of the subjects did not receive another arthroplasty within the study period. Between 16% and 22% of subjects who
had a first hip or knee arthroplasty received a second hip or knee arthroplasty respectively. If the second arthroplasty was in the same type of joint (cognate joint) as the first, only 1-2% went on to have a third arthroplasty. If the second arthroplasty was in a non-cognate joint (e.g. hip followed by knee), 9-17% of subjects went on to have another arthroplasty in the contralateral joint to the second arthroplasty (state $3 \rightarrow 8$ and state $4 \rightarrow 8$). The highlighted transitions in Table 7.2 correspond to the HRs presented in Table 7.3.

Table 7.3 shows the effect of side of first arthroplasty (either hip or knee) adjusted for age and sex on the transition hazards between the states highlighted in Figure 7.1 and Table 7.2. Occurrence of revision was included in transitions where it had a significant effect. After the first hip arthroplasty the hazard of receiving a knee on the contralateral side was higher than the hazard of receiving a knee on the ipsilateral side. That is, for subjects who had received a hip first, the hazard ratio (right vs. left first hip) of receiving a left knee (state $1 \rightarrow 3$) was 1.83 (95% confidence interval (CI): 1.65, 2.02) and 2.97 (95% CI: 2.10, 4.20) for Australian and Norwegian subjects respectively (illustrated in Figure 7.2), whereas for receiving a right knee (state $1 \rightarrow 4$) the hazard ratio was 0.52 (95% CI: 0.48, 0.57) for Australians and 0.51 (95% CI: 0.40, 0.65) for Norwegians.

For subjects who received a second knee following a hip and a knee (state $3 \rightarrow 8$ and state $4 \rightarrow 8$), there was a higher hazard of receiving a knee contralateral than ipsilateral to the index hip. That is, the hazard ratio (right vs. left first hip) of receiving a subsequent right knee (state $3 \rightarrow 8$) was 0.74 (95% CI: 0.56, 0.99) and 0.72 (95% CI: 0.26, 1.95) for Australian and Norwegian subjects respectively (illustrated in Figure 7.3), whereas for receiving a subsequent left knee (state $-4 \approx 8$) the ha zard ratio was 1.62 (95% CI: 1.26, 2.10) for Australians and 2.09 (95% CI: 1.01, 4.35) for Norwegians. For subjects who received a knee as a first arthroplasty evidence of a pattern was less consistent than for subjects who received a hip first (Table 7.3). The transition hazard of receiving a third arthroplasty after two arthroplasties of the same

type of joint (e.g. after bilateral hip arthroplasties) did not show a consistent association with the side of the first arthroplasty (Table 7.3, state $2 \rightarrow 6$ and state $2 \rightarrow 7$).

Figures 7.4 and 7.5 show the estimated probabilities of occupying state 3 and 4 (Figure 7.1) over a period of 5 years after the first arthroplasty. Figure 7.4 shows the estimated probabilities for individuals who had received a hip arthroplasty followed by a knee arthroplasty (state 3 on the left panel and state 4 on the right panel). The figure indicates that from approximately half a year after the initial arthroplasty the probabilities of having received a contralateral knee were consistently higher than the probabilities of having received an ipsilateral knee. For example, after 5 years the estimated probabilities of having received a left knee (contralateral) after a right hip were approximately 2.9 % and 1.5% for Australians and Norwegians respectively, whereas the probabilities of having received a left knee (ipsilateral) after a left hip were approximately 1.5% and 0.4% respectively (left panel Figure 7.4). Figure 7.5 shows the estimated probabilities for individuals who had received a knee arthroplasty followed by a hip arthroplasty (state 3 on the left panel and state 4 on the right panel). For both countries, there was less difference in the probabilities between receiving a contralateral hip and an ipsilateral hip. For example, after 5 years the estimated probabilities of having received a left hip (contralateral) after a right knee were approximately 1.2 % and 1.1% for Australians and Norwegians respectively. The probabilities of having received a left hip (ipsilateral) after a left knee were approximately 0.9% and 1.1% for Australians and Norwegians respectively. The probabilities of having received a left hip after a knee arthroplasty were also similar for the two countries, but the probabilities of having received a right hip after a knee was somewhat higher for the Norwegians than Australians over the 5-year period.

 Table 7.3: Effect of side of first arthroplasty (hip or knee) on hazards for selected transitions in the model.

	Right vs. left hip		Right vs. left knee	
	Australia	Norway	Australia	Norway
	HR (95% CI)	HR (95% CI)	HR (95% CI)	HR (95% CI)
State 1→3	1.83 (1.65, 2.02)***	2.97 (2.10, 4.20)***	1.10 (0.99, 1.23)	1.08 (0.76, 1.52)
1→4	0.52 (0.48, 0.57)***	0.51 (0.40, 0.65)***	0.87 (0.79, 0.95)*	0.77 (0.59, 1.001)
State 2→6	1.25 (0.92, 1.69)	1.12 (0.52, 2.46)	0.95 (0.74, 1.20)	1.36 (0.63, 2.93)
2 →7	0.74 (0.58, 0.96)*	0.97 (0.54, 1.74)	1.05 (0.85, 1.30)	0.83 (0.47, 1.49)
State 3→8	0.74 (0.56, 0.99)*	0.72 (0.26, 1.95)	0.87 (0.65, 1.22)	0.64 (0.26. 1.53)
State 4→8	1.62 (1.26,2.10)***	2.09 (1.01, 4.35)*	1.34 (0.98, 1.84)	0.97 (0.45, 2.08)

***P<0.001; *P<0.05. HR: Hazard ratio



Figure 7.2: Comparing hazards of receiving a left knee arthroplasty between individuals who had received a right hip arthroplasty with individuals who had received a left hip arthroplasty. HR: hazard ratio, $\lambda(t|R)_{1\rightarrow3}$: hazard of receiving a left knee given that first hip was a right hip, $\lambda(t|L)_{1\rightarrow3}$: hazard of receiving a left knee given that first hip was a left hip.



Figure 7.3: Comparing hazards of receiving a right knee between individuals who had received arthroplasties in right hip and left knee with individuals who had received arthroplasties in left hip and left knee. HR: hazard ratio, $\lambda(t|R)_{3\rightarrow8}$: hazard of receiving a right knee given that first hip was a right hip, $\lambda(t|L)_{3\rightarrow8}$: hazard of receiving a right knee given that first hip was a left hip.



Figure 7.4: Estimated probabilities for receiving a knee arthroplasty after having received a hip arthroplasty (AU: Australia, NOR: Norway, left panel: state 3, right panel: state 4).



Figure 7.5: Estimated probabilities for receiving a hip arthroplasty after having received a knee arthroplasty (AU: Australia, NOR: Norway, left panel: state 3, right panel: state 4).

7.3.5 Discussion

We used a multi-state model to investigate the progression of joint replacements in large weight bearing joints. The majority of individuals who received a second arthroplasty did so in the cognate contralateral joint. If the first arthroplasty was a hip, the hazards of receiving subsequent knee arthroplasties were higher on the contralateral side than on the ipsilateral side to the index arthroplasty. This was most pronounced for progression from hip arthroplasty to first knee arthroplasty but was also evident on the transition to receiving another (second) knee. Hence the side of first hip arthroplasty affected the rate of receiving subsequent knee arthroplasties. If the first arthroplasty was a knee, the hazards of receiving subsequent hip arthroplasties were generally higher on the contralateral side, but were not statistically significant.

The estimated HRs express the relative effect of side of the first arthroplasty on the subsequent transitions to arthroplasties in other joints, but not the absolute probabilities of individuals receiving further arthroplasties. We therefore also estimated probabilities for transitions from the first hip or knee arthroplasty to a second arthroplasty in a noncognate joint. They showed the same pattern as the estimated HRs for the respective transitions. After 5 years, the probabilities of having received a knee contralateral to the index hip was more than 1.7 times the probabilities of having received a knee ipsilateral to the index hip, whereas there was little difference between the sides of hips relative to the previous knee arthroplasties. This was evident in data from both countries, but the probabilities for Australians to have received a knee after a hip were higher than for Norwegians (Figure 7.4). Australia has a higher incidence of knee arthroplasties than Norway [9], and the difference in probabilities may be partially explained by this. The somewhat higher probabilities for Norwegians compared to Australians to receive a right hip (but not left hip) after a knee arthroplasty are difficult to explain. They may be related to particular risk factors for OA in the Norwegian population as it has previously been reported that Norway has much higher incidence ratios (female/male) for THA than the other Nordic countries [145].

Our study has some limitations. Joint replacements may not be an ideal measure of the incidence of symptomatic end-stage OA. Several factors contribute to regional and national variation in rate of surgical treatment for OA, such as access to treatment, disparity by race or ethnic group, surgical waiting lists, socio-economic status, patient or orthopaedic surgeon preferences [146-149]. However, our study sample consisted of individuals who had received one arthroplasty for OA and some of the above factors would likely have less influence since the subjects already had been selected once for the same treatment. Furthermore, although there were differences in the relative proportion of hip and knee replacements in the Australian and the Norwegian registry data, the pattern of subsequent arthroplasties were similar between the two countries and are therefore likely to reflect the sequence of OA progression.

The results are consistent with those of other studies that have shown the progression of end-stage OA in large weight bearing joints to be a non random process [129, 150, 151], but we did not find clear evidence of a difference in association between side of the first knee arthroplasty and the following hip arthroplasty as has been described by others [129]. This may be due to differences in study design. Our use of the multi-state modelling technique allows for a more comprehensive analysis of the data than previous studies. It enables the analysis of the entire arthroplasty history of interest which in our study was the sequence and timing of joint replacements after the first hip or knee to subsequent hip(s)/knee(s). The multi-state model, which is a generalisation of standard survival analysis of time to one event, not only takes the time to different events into account but also incorporates incomplete observations, that is, the information contained in the time that some subjects have been under observation without experiencing the event(s). Another strength of this study is that it is the first study using data from two large, independent population-based national arthroplasty registries showing that there is evidence for a pattern in the progression of OA. Previous studies have involved far fewer subjects, from 50 to 3000, compared to our study, which entails more than 230,000 subjects. Both registries have excellent coverage of joint replacement procedures performed in the respective countries[81, 143] and the two countries have developed health systems [152]. The Australian registry is comparatively large whereas the Norwegian registry has been operating for more than 20 years, thus data from the two registries complement each other.

Several studies have found evidence of bilateral symmetrical OA in large weight bearing joints [153] which may indicate that some individuals have an increased susceptibility to develop OA, due to systemic factors and/or genetic factors. In addition, biomechanical factors may contribute to the progression of end-stage OA to the contralateral joint. Shakoor et al.[154] performed a gait study in 62 patients with unilateral symptomatic and radiological hip OA, and found evidence that the contralateral asymptomatic knee and hip had increased dynamic loading as well as increased medial compartment tibial bone mineral density. Hence, the consequence of gait alterations due to a diseased hip may be responsible for the subsequent development of OA in the contralateral knee. Furthermore, several studies have found that gait does not return to normal after THA and TKA [155-159] which may explain the progression of OA in the non-operated limbs. Umeda et al. [150] did a longitudinal study in 30 women who had received hip arthroplasty, most for developmental dysplasia. Baseline radiographs showed no difference in knee OA between the operated and the non-operated side. At follow up, after minimum 10 years, there was significantly more severe knee OA medially in the non-THA side than the THA side. The authors concluded that this could be related to reduced offset of conventional femoral prostheses leading to shifts in mechanical axes. Further, leg length discrepancy (LLD) after THA is common [160]. Tanaka et al. [161] found that postoperative LLD and stage of preoperative hip OA were the factors that had the largest influence on gait abnormalities after THA. Hence, the pattern of progression of joint replacements in large weight bearing joints, especially after the first hip replacement, may be related to LLD and associated pre and/or postoperative gait abnormalities. This is consistent with the work of Harvey et al. [162] who found that LLD was associated with prevalence, incidence and progression of knee OA. However, the registries have no access to data on LLD so we can make no definite statement about this. Further studies are needed to investigate whether the increased risk of receiving an arthroplasty in the contralateral

knee is related to having a THA and/or preoperative factors such as pain and altered gait associated with hip OA.

In conclusion, we have demonstrated in data from two large population-based national arthroplasty registries of 55-74 year old subjects who received arthroplasties for OA, that there is evidence of an association between the side of the first hip arthroplasty and side of subsequent knee arthroplasties. This is indicative of a pattern of progression of OA in large weight bearing joints. The evidence of a pattern in the progression of joint replacements and the nature of this pattern are important for the understanding of the pathogenesis of OA as well as for prevention and treatment.

END OF PUBLISHED ARTICLE

8 SUMMARY AND CONCLUSIONS

In this thesis statistical methods for analysing joint replacement registry data have been investigated. Joint replacement registries are important sources of data for evaluating outcomes of joint replacements. The use of appropriate statistical methods to analyse the data is of great importance. The overall aim of the thesis was to explore methods for analysing time to event data contained in joint replacement registries. The emphasis has been on multiple event data, more specifically methods for competing risks scenarios and multi-state modelling.

In this final chapter, I summarise the main findings and contributions to knowledge, discuss limitations and address topics for future research.

8.1 Main findings and contributions

8.1.1 Non-parametric competing risks methods and arthroplasty data

The Kaplan-Meier (KM) method is generally known to overestimate the probability of failure in the presence of competing risks. An important aim of joint registries is to identify poorly performing prostheses. Revision surgery is considered the main indicator of failure of the prosthesis. In Chapter 4, the aim was to apply KM methods to arthroplasty data and compare the estimates of risk of revision with estimates accounting for the competing risk of death. In order to evaluate the magnitude of the potential bias when using the KM method, subsets of data with different incidence of the competing risk of death were examined.

It was evident that when the risk of the competing event death was high, such as that in individuals who received arthroplasty for fractured neck of femur, the KM method substantially overestimated the risk of revision. For example, 5 years after insertion of the primary procedure, the relative difference between the KM and CIF estimates was 79% for the group of patients with the highest mortality and 1.4% for the group with the

lowest mortality. The relative difference increased with time due to the incidence of death increasing with increasing age.

The results from this thesis are important since survival (or failure) curves are often used to estimate the risk of revision of prostheses over time, and as this work demonstrates, these estimates may be substantially biased if competing risks are not taken into account. Scenarios where accounting for competing risks in joint replacement registries are especially relevant include those where the incidence of death is high, where one reason for revision competes with other reasons for revision, or where there are long follow-up times. Prior to publication of the article in Chapter 4, only a few studies in orthopaedic research had been published addressing competing risks. This was the first article to show evidence of clinically relevant bias in the KM method on data from a large joint replacement registry. The article makes recommendations for analysis of joint replacement data and informs clinicians and other readers on how to interpret results from the KM method published in registry reports.

8.1.2 Competing risks regression and arthroplasty data

Joint replacement registries collect and record information on a number of characteristics of the recipients of joint prostheses such as age and sex, type of prosthesis, type of fixation, indication for the procedure etc. In the analysis of these data, it is of interest to obtain estimates of effects on hazards and risks of failures adjusted for appropriate covariates. Several approaches for regression modelling in competing risks scenarios exist, but prior to the article presented in Chapter 5, their use had not been examined in the analysis of joint replacement registry data. The aim of the article in Chapter 5 was to describe some of the available models for dealing with competing risks in the analysis of joint replacement registry data and use them to examine the effects of covariates on the hazard rates and the risks of revision.

The effects of various covariates on the hazard of revision were modelled with a Cox-Aalen model, allowing the effect of fixation (cementless vs. cemented) to vary with time. The effect on the actual risk of revision, treating death as a competing risk, was modelled with an extension of the Fine and Gray model, also allowing the effect of fixation to vary with time. The key finding in the article was that some covariates had a different effect on the hazard of revision than on the actual risk of revision, thus demonstrating that in competing risks scenarios, the effect on the risk of an event depends on the cause specific hazard for every event. Further, predictions based on both models showed that estimates based on the Cox-Aalen model overestimated the risk of revision compared to estimates from the modified Fine and Gray model. This was most pronounced for the group with the highest incidence of the competing risk of death.

The article in Chapter 5 is the first to demonstrate that the effect of covariates on the hazard rate and risk of revision may be different. Thus, as discussed in Chapter 4, results from joint replacement registries must be interpreted with care when competing risks are present. The articles in Chapters 4 and 5 should contribute to increased awareness of competing risks issues in the orthopaedic research and registry community and alert analysts to use appropriate tools in the presence of competing risks. Furthermore, although the Cox PH model is the standard regression model in time to event analysis, Aalen's additive model has many advantages as demonstrated in the article in Chapter 5 where the changing effect of a time-varying covariate can be visualised and thereby provide a clearer understanding of the data.

8.1.3 Multi-state models and arthroplasty histories

With the increases in life expectancy of the population, the increase in the rate of joint replacements is predicted to continue, meaning also that patients are likely to have several joint replacement procedures during their lifetime. Thus, data contained in joint replacement registries is expanding and is getting more complex. The aim of the article in Chapter 6 was to examine the use of multi-state modelling techniques in the analysis of data on complex arthroplasty histories with several outcomes of interest occurring over time. The aim was also to develop a system of notation that could assist in the management and analysis of large joint replacement data sets.

A multi-state model was developed where individuals were followed as they moved through different possible states from their first hip arthroplasty to receiving a second arthroplasty, to revision, death or until study closure. The model presented in Chapter 6 was used to describe numbers and proportions of patients in the cohort who experienced these events, their transition probabilities and the effect of sex on the transition intensities in the model. It was demonstrated that multi-state modelling techniques were well suited to the analysis and description of complex arthroplasty histories. The Summary Notation for Arthroplasty Histories (SNAH) code proved to be extremely useful when managing the large and complex data set.

The article in Chapter 6 contributes to the increasing research literature on the application and use of multi-state modelling techniques in medical research. The development has been facilitated by more advanced computer capabilities and increased availability of necessary software. The work presented in Chapter 6 will lead to increased awareness of advantages and possibilities of multi-state models in the analysis of multiple events amongst researchers using arthroplasty data. No system for managing complex arthroplasty histories has existed until now. The SNAH code could prove useful not only in the management of arthroplasty data, but also in a clinical context describing a patient's arthroplasty history.

8.1.4 Application of multi-state models and osteoarthritis

Data from joint replacement registries contain unique information on joint replacements in the population. For example, the indication for patients receiving an arthroplasty is recorded. Joint replacement is often a successful treatment for end-stage osteoarthritis and symptomatic osteoarthritis is the most common indication for joint replacement. Subsequently, the incidence of joint replacement is considered an indicator of the incidence of end stage osteoarthritis. The aim of the article in Chapter 7 was to apply multi-state modelling methods to joint replacement registry data to investigate progression of hip and knee osteoarthritis. Data were obtained from two national joint replacement registries. A multi-state model was developed where individuals were followed from a first hip or knee arthroplasty for osteoarthritis to subsequent hip or knee arthroplasties, death or study closure. The effect of side of the first arthroplasty on the transition intensities and transition probabilities was estimated. The article demonstrated that there was evidence of an association between the side of the first hip arthroplasty and the side of subsequent knee arthroplasty.

Osteoarthritis is a common chronic disease. Factors such as older age and obesity are known to increase the risk, but the pathogenesis of osteoarthritis is not clear. Results from the article in Chapter 7 contribute to an understanding of the pathogenesis of osteoarthritis informed by the sequence of progression of osteoarthritis to different joints. This has implications for further research into causes, prevention and treatment of osteoarthritis. This article also illustrates the application of multi-state models to this important area of research and contributes information about alternative approaches to the analysis and use of arthroplasty registry data.

8.2 Limitations and future directions

Analysis of time to event data is a research area with a high level of activity. This thesis has focused on multi-state modelling to explore event histories contained in arthroplasty data, both in competing risks scenarios and when there are multiple joint procedures per patient. There are other approaches to analysing time to event data with multiple events. It would be of interest to examine how multi-state models compare with other models, such as marginal and conditional models in the analysis of arthroplasty data. Although the role of frailties in multi-state models is not clear [163], future work may consider their use accounting for clustering of events in arthroplasty data, for example by hospitals. Further, in this thesis, data on patients who received two arthroplasties on the same day were excluded and the performance of different types of models to analyse multivariate event data such as these would be of interest.

The most commonly used methods to analyse arthroplasty registry data are the Kaplan-Meier survival curves and the Cox PH model. Further investigation into the use of alternatives to the Cox PH model is warranted. An extension of one of these alternatives, Aalen's additive model, was used in this thesis and Aalen's additive model should be used more widely in the analysis of time to event data. The National Joint Registry in the United Kingdom, which is the largest joint replacement registry in the world, use a flexible parametric model to obtain relative and absolute hazards as well as to account for time varying effect and competing risks. Comparisons of this approach with the multi-state model used in this thesis would be interesting as both approaches have theoretical advantages over other (more commonly utilised) models. A related issue of interest is collaboration between national registries in evaluating statistical methods to analyse the registry data. Although national joint replacement registries operate mainly for their own use and have national characteristics [164], standardisation in reporting and analysis methods would be useful in disseminating and comparing results amongst stakeholders in registries worldwide. Continued evaluation of statistical methods to analyse the data presented in annual reports is of utmost importance. Because national regulations differ, various restrictions exist in gaining access to data in national joint replacements registries [71]. The data for this thesis were raw deidentified data obtained from the Australian and Norwegian registries offering a unique opportunity to examine characteristics of both registries. Continued collaboration of this kind is important and sharing of information across countries is beneficial for all parties involved.

In many joint replacement registries such as the AOA NJRR and the NAR, the amount of data collected on each joint replacement procedure is relatively small which contributes to good compliance and reporting. However, the resulting outcome measure, such as revision, consequently becomes crude as it may depend on many factors that are not recorded. One way to handle this issue, which will become increasingly important, is to link joint replacement data to other types of registries and data bases. Some of these other data bases may contain important measurements that are continually updated. An interesting approach advocated by Aalen [165], makes use of these additional data by integrating time to event and longitudinal measurement data in the analysis and may prove to be a very useful tool in describing complex arthroplasty histories.

The SNAH code described in the article in Chapter 6 was developed to describe and manage arthroplasty history event data. Future developments could consider a similar approach for the notation of sequential events in other types of registry data where there is interest in analysing and keeping track of patients' event histories. For example, outcomes of cancer treatment are often modelled with a multi-state model and a similar summary notation could be used to describe the sequence of treatments and relapses after the diagnosis of cancer for each individual patient.

8.3 Conclusion

This thesis has investigated methods for analysing time to event data contained in joint replacement registry data. The focus has been on exploring how multi-state models can be used to handle multiple events in these data. It has been demonstrated that in the presence of a competing risk such as death, the Kaplan-Meier method might lead to clinically relevant biased estimates of the risk of revision. The effect of covariates on the risk and on the rate of revision may differ and it is important that estimates obtained from the Cox PH model and competing risks regression models are interpreted correctly. Multi-state models provide a useful tool to describe and analyse data containing complex arthroplasty histories.

9 REFERENCES

- 1. Gliklich, R.E. and N.A. Dreyer, eds. *Registries for Evaluating Patient Outcomes: A User's Guide*. 2nd edition ed. 2010, Rockville (MD): Agency for Healthcare Research and Quality (US).
- 2. Hershman, D.L. and J.D. Wright, *Comparative effectiveness research in oncology methodology: observational data.* J Clin Oncol, 2012. **30**(34): p. 4215-22.
- 3. Vandenbroucke, J.P., *Observational research, randomised trials, and two views of medical science.* PLoS Med, 2008. **5**(3): p. e67.
- 4. Concato, J., N. Shah, and R.I. Horwitz, *Randomized, controlled trials, observational studies, and the hierarchy of research designs.* N Engl J Med, 2000. **342**(25): p. 1887-92.
- 5. Hoppe, D.J., et al., *Hierarchy of evidence: where observational studies fit in and why we need them.* J Bone Joint Surg Am, 2009. **91 Suppl 3**: p. 2-9.
- 6. Silverman, S.L., *From randomized controlled trials to observational studies*. Am J Med, 2009. **122**(2): p. 114-20.
- 7. Labek, G., et al., Organisation, data evaluation, interpretation and effect of arthroplasty register data on the outcome in terms of revision rate in total hip arthroplasty. Int Orthop, 2011. **35**(2): p. 157-63.
- 8. Sedrakyan, A., et al., *The International Consortium of Orthopaedic Registries: overview and summary*. J Bone Joint Surg Am, 2011. **93 Suppl 3**: p. 1-12.
- 9. Robertsson, O., et al., *Knee arthroplasty in Denmark, Norway and Sweden*. Acta Orthop, 2010. **81**(1): p. 82-9.
- 10. Labek, G., C.I. Stoica, and N. Bohler, *Comparison of the information in arthroplasty registers from different countries.* J Bone Joint Surg Br, 2008. **90**(3): p. 288-91.
- 11. von Knoch, F., A. Marchie, and H. Malchau, *Total joint registries: a foundation for evidence-based arthroplasty.* Virtual Mentor, 2010. **12**(2): p. 124-9.
- 12. Kurtz, S., et al., *Projections of primary and revision hip and knee arthroplasty in the United States from 2005 to 2030.* J Bone Joint Surg Am, 2007. **89**(4): p. 780-5.

- 13. AOANJRR, Australian Orthopaedic Association National Joint Replacement Registry, Annual Report, 2012: Adelaide.
- 14. Kaplan, E.L. and P. Meier, *Non-parametric estimation from incomplete observations*. Journal of the American Statistical Association, 1958. **53**: p. 457-481.
- 15. Cox, D.R., *Regression models and life tables (with discussion)*. Journal of the Royal Statististical Society, Series B, 1972. **34**: p. 187- 220.
- 16. Ryan, T. and W. Woodall, *The most-cited statistical papers*. Journal of Applied Statistics, 2005. **32**(5): p. 461-474.
- 17. Koller, M.T., et al., *Competing risks and the clinical community: irrelevance or ignorance?* Stat Med, 2011.
- 18. Aalen, O.O., *Medical statistics--no time for complacency*. Stat Methods Med Res, 2000. **9**(1): p. 31-40.
- 19. Schemper, M., S. Wakounig, and G. Heinze, *The estimation of average hazard ratios by weighted Cox regression*. Stat Med, 2009. **28**(19): p. 2473-89.
- 20. Aalen, O.O., *A linear regression model for the analysis of life times.* Stat Med, 1989. **8**(8): p. 907-25.
- 21. Fine, J.P. and R.J. Gray, *A proportional hazards model for the subdistribution of a competing risk.* Journal of the American Statistical Association, 1999. **94**(446): p. 496-509.
- 22. Andersen, P.K. and N. Keiding, *Interpretability and importance of functionals in competing risks and multistate models*. Stat Med, 2012. **31**(11-12): p. 1074-88.
- 23. Mantel, N., *Evaluation of survival data and two new rank order statistics arising in its consideration*. Cancer Chemother Rep, 1966. **50**(3): p. 163-70.
- 24. Fleming, T.R. and D.Y. Lin, *Survival analysis in clinical trials: past developments and future directions.* Biometrics, 2000. **56**(4): p. 971-83.
- 25. Nelson, W., *Theory and applications of hazard plotting for censored failure data*. Technometrics, 2000. **42**(1): p. 12-25.
- 26. Aalen, O.O., *Nonparametric Inference for a Family of Counting Processes*. Annals of Statistics, 1978. **6**(4): p. 701-726.

- 27. Hougaard, P., *Analysis of multivariate survival data*. Statistics for biology and health. 2000, New York: Springer. xvii, 542 p.
- 28. Andersen, P.K., et al., *Competing risks in epidemiology: possibilities and pitfalls.* Int J Epidemiol, 2012. **41**(3): p. 861-70.
- 29. Cooke, R.M. and O. Morales-Napoles, *Competing risk and the Cox proportional hazard model*. Journal of Statistical Planning and Inference, 2006. **136**(5): p. 1621-1637.
- Kalbfleisch, J.D. and R.L. Prentice, *The statistical analysis of failure time data*.
 2nd ed. Wiley series in probability and statistics. 2002, Hoboken, N.J.: J. Wiley.
 xiii, 439 p.
- 31. Cortese, G. and P.K. Andersen, *Competing risks and time-dependent covariates*. Biom J, 2010. **52**(1): p. 138-58.
- van Walraven, C., et al., *Time-dependent bias was common in survival analyses published in leading clinical journals.* Journal of Clinical Epidemiology, 2004. 57(7): p. 672-682.
- 33. Beyersmann, J., M. Wolkewitz, and M. Schumacher, *The impact of time-dependent bias in proportional hazards modelling*. Stat Med, 2008. **27**(30): p. 6439-54.
- 34. Fisher, L.D. and D.Y. Lin, *Time-dependent covariates in the Cox proportionalhazards regression model.* Annu Rev Public Health, 1999. **20**: p. 145-57.
- 35. Bellera, C.A., et al., Variables with time-varying effects and the Cox model: some statistical concepts illustrated with a prognostic factor study in breast cancer. BMC Med Res Methodol, 2010. **10**: p. http://www.biomedcentral.com/1471-2288/10/20.
- 36. Buchholz, A. and W. Sauerbrei, *Comparison of procedures to assess non-linear* and time-varying effects in multivariable models for survival data. Biom J, 2011. **53**(2): p. 308-31.
- 37. Aalen, O.O., Ø. Borgan, and S. Gjessing, *Survival and event history analysis : a process point of view*. Statistics for biology and health. 2008, New York, NY: Springer. xviii, 539 p.
- 38. Bradburn, M.J., et al., Survival analysis part II: multivariate data analysis--an introduction to concepts and methods. Br J Cancer, 2003. **89**(3): p. 431-6.

- 39. Lee, E.W., L.J. Wei, and D.A. Amato, *Cox-type regression analysis for large numbers of small groups of correlated failure time observations*. Survival analysis : state of the art, NATO ASI series. Series E, Applied sciences, ed. J.P. Klein and P.K. Goel. 1992, Dordrecht ; Boston: Kluwer Academic Publishers. x, 451 p.
- 40. Wei, L.J., D.Y. Lin, and L. Weissfeld, *Regression-Analysis of Multivariate Incomplete Failure Time Data by Modeling Marginal Distributions*. Journal of the American Statistical Association, 1989. **84**(408): p. 1065-1073.
- 41. Andersen, P.K. and R.D. Gill, *Cox Regression-Model for Counting-Processes a Large Sample Study*. Annals of Statistics, 1982. **10**(4): p. 1100-1120.
- 42. Prentice, R.L., B.J. Williams, and A.V. Peterson, *On the Regression-Analysis of Multivariate Failure Time Data*. Biometrika, 1981. **68**(2): p. 373-379.
- 43. Vaupel, J.W., K.G. Manton, and E. Stallard, *The impact of heterogeneity in individual frailty on the dynamics of mortality*. Demography, 1979. **16**(3): p. 439-54.
- 44. Therneau, T.M. and P.M. Grambsch, *Modeling survival data : extending the Cox model*. Statistics for biology and health. 2000, New York: Springer. xiii, 350 p.
- 45. Klein, J.P. and Y. Shu, *Multi-state models for bone marrow transplantation studies*. Stat Methods Med Res, 2002. **11**(2): p. 117-39.
- 46. Andersen, P.K., et al., *Statistical models based on counting processes*. Springer series in statistics. 1993, New York: Springer-Verlag. xi, 767 p.
- 47. Commenges, D., *Multi-state models in epidemiology*. Lifetime Data Anal, 1999. **5**(4): p. 315-27.
- 48. Meira-Machado, L., et al., *Multi-state models for the analysis of time-to-event data*. Stat Methods Med Res, 2009. **18**(2): p. 195-222.
- 49. Andersen, P.K. and N. Keiding, *Multi-state models for event history analysis*. Stat Methods Med Res, 2002. **11**(2): p. 91-115.
- 50. Hougaard, P., *Multi-state models: a review*. Lifetime Data Anal, 1999. **5**(3): p. 239-64.
- 51. Putter, H., M. Fiocco, and R.B. Geskus, *Tutorial in biostatistics: competing risks and multi-state models.* Stat Med, 2007. **26**(11): p. 2389-430.

- 52. Andersen, P.K. and M. Pohar Perme, *Inference for outcome probabilities in multi-state models*. Lifetime Data Anal, 2008. **14**(4): p. 405-31.
- 53. Datta, S. and G.A. Satten, Validity of the Aalen-Johansen estimators of stage occupation probabilities and Nelson-Aalen estimators of integrated transition hazards for non-Markov models. Statistics & Probability Letters, 2001. 55(4): p. 403-411.
- 54. Gunnes, N., O. Borgan, and O.O. Aalen, *Estimating stage occupation* probabilities in non-Markov models. Lifetime Data Anal, 2007. **13**(2): p. 211-40.
- 55. Gooley, T.A., et al., *Estimation of failure probabilities in the presence of competing risks: new representations of old estimators.* Stat Med, 1999. **18**(6): p. 695-706.
- 56. Crowder, M.J., *Classical competing risks*. 2001, Boca Raton: Chapman & Hall/CRC. 186 p.
- 57. Dietz, K. and J.A. Heesterbeek, *Daniel Bernoulli's epidemiological model revisited*. Math Biosci, 2002. **180**: p. 1-21.
- 58. Andersen, P.K., S.Z. Abildstrom, and S. Rosthoj, *Competing risks as a multi-state model*. Stat Methods Med Res, 2002. **11**(2): p. 203-15.
- 59. Aalen, O.O. and S. Johansen, *Empirical Transition Matrix for Nonhomogeneous Markov-Chains Based on Censored Observations*. Scandinavian Journal of Statistics, 1978. **5**(3): p. 141-150.
- 60. Lunn, M. and D. McNeil, *Applying Cox regression to competing risks*. Biometrics, 1995. **51**(2): p. 524-32.
- 61. Tai, B.C., et al., *Competing risks analysis of patients with osteosarcoma: a comparison of four different approaches.* Stat Med, 2001. **20**(5): p. 661-84.
- 62. Zhang, M.J. and J. Fine, *Summarizing differences in cumulative incidence functions*. Stat Med, 2008. **27**(24): p. 4939-49.
- 63. Klein, J.P. and P.K. Andersen, *Regression modeling of competing risks data* based on pseudovalues of the cumulative incidence function. Biometrics, 2005. **61**(1): p. 223-9.
- 64. Andersen, P.K. and M.P. Perme, *Pseudo-observations in survival analysis*. Stat Methods Med Res, 2010. **19**(1): p. 71-99.

- 65. Gray, R.J., A Class of K-Sample Tests for Comparing the Cumulative Incidence of a Competing Risk. Annals of Statistics, 1988. **16**(3): p. 1141-1154.
- 66. Marashi-Pour, S., et al., *Competing risk analysis of mortality from invasive cutaneous melanoma in New South Wales: a population-based study, 1988-2007.* Aust N Z J Public Health, 2012. **36**(5): p. 441-5.
- 67. Omsland, T.K., et al., *Ten-year risk of second hip fracture. A NOREPOS study.* Bone, 2013. **52**(1): p. 493-7.
- 68. Kim, H.T., *Cumulative incidence in competing risks data and competing risks regression analysis.* Clin Cancer Res, 2007. **13**(2 Pt 1): p. 559-65.
- 69. Zhang, M.J., X. Zhang, and T.H. Scheike, *Modeling cumulative incidence function for competing risks data*. Expert Rev Clin Pharmacol, 2008. **1**(3): p. 391-400.
- 70. Haller, B., G. Schmidt, and K. Ulm, *Applying competing risks regression models: an overview*. Lifetime Data Anal, 2012.
- 71. Boyer, P., I. Boutron, and P. Ravaud, *Scientific production and impact of national registers: the example of orthopaedic national registers*. Osteoarthritis Cartilage, 2011. **19**(7): p. 858-63.
- 72. Kolling, C., et al., *Key factors for a successful National Arthroplasty Register*. J Bone Joint Surg Br, 2007. **89**(12): p. 1567-73.
- 73. Robertsson, O., *Knee arthroplasty registers*. J Bone Joint Surg Br, 2007. **89**(1): p. 1-4.
- 74. NAR, The Norwegian Arthroplasty Register Annual Report. 2012.
- 75. Ranstam, J., et al., *Statistical analysis of arthroplasty data*. I. Introduction and *background*. Acta Orthop, 2011. **82**(3): p. 253-7.
- 76. Royston, P. and M.K. Parmar, *Flexible parametric proportional-hazards and proportional-odds models for censored survival data, with application to prognostic modelling and estimation of treatment effects.* Stat Med, 2002. **21**(15): p. 2175-97.
- Smith, A.J., et al., Failure rates of metal-on-metal hip resurfacings: analysis of data from the National Joint Registry for England and Wales. Lancet, 2012. 380(9855): p. 1759-66.

- 78. Labek, G., et al., *Revision rates after total joint replacement: cumulative results from worldwide joint register datasets.* J Bone Joint Surg Br, 2011. **93**(3): p. 293-7.
- 79. Havelin, L.I., et al., *The Norwegian Arthroplasty Register: 11 years and 73,000 arthroplasties.* Acta Orthop Scand, 2000. **71**(4): p. 337-53.
- 80. NAR. *The Norwegian Arthroplasty Register*. 2012 20 December 2012]; Available from: <u>http://nrlweb.ihelse.net/eng/default.htm</u>.
- 81. Espehaug, B., et al., *Registration completeness in the Norwegian Arthroplasty Register*. Acta Orthop, 2006. **77**(1): p. 49-56.
- 82. Espehaug, B., et al., *Hip arthroplasty in Norway 1989-2008*. Tidsskr Nor Laegeforen, 2011. **131**(16): p. 1543-8.
- 83. Gillam, M.H., et al., *Competing risks survival analysis applied to data from the Australian Orthopaedic Association National Joint Replacement Registry*. Acta Orthop, 2010. **81**(5): p. 548-55.
- 84. AOANJRR, Australian Orthopaedic Association National Joint Replacement Registry, Annual Report, 2009, AOA: Adelaide.
- 85. Prentice, R.L., et al., *The analysis of failure times in the presence of competing risks*. Biometrics, 1978. **34**(4): p. 541-54.
- 86. Kalbfleisch, J.D. and R.L. Prentice, *The statistical analysis of failure time data*. Wiley series in probability and mathematical statistics. 1980, New York: Wiley. xi, 321 p.
- 87. Biau, D.J., A. Latouche, and R. Porcher, *Competing events influence estimated survival probability: when is Kaplan-Meier analysis appropriate?* Clin Orthop Relat Res, 2007. **462**: p. 229-33.
- 88. Dignam, J.J. and M.N. Kocherginsky, *Choice and interpretation of statistical tests used when competing risks are present.* J Clin Oncol, 2008. **26**(24): p. 4027-34.
- Pocock, S.J., T.C. Clayton, and D.G. Altman, *Survival plots of time-to-event outcomes in clinical trials: good practice and pitfalls*. Lancet, 2002. 359(9318): p. 1686-9.
- 90. Schwarzer, G., et al., *Statistical analysis of failure times in total joint replacement*. J Clin Epidemiol, 2001. **54**(10): p. 997-1003.

- 91. Ranstam, J. and O. Robertsson, *Statistical analysis of arthroplasty register data*. Acta Orthop, 2010. **81**(1): p. 10-4.
- 92. Fennema, P. and J. Lubsen, Survival analysis in total joint replacement: an alternative method of accounting for the presence of competing risk. J Bone Joint Surg Br, 2010. **92**(5): p. 701-6.
- 93. Kim, W.R., et al., *Deaths on the liver transplant waiting list: an analysis of competing risks.* Hepatology, 2006. **43**(2): p. 345-51.
- 94. Resche-Rigon, M., E. Azoulay, and S. Chevret, *Evaluating mortality in intensive care units: contribution of competing risks analyses.* Crit Care, 2006. **10**(1): p. R5.
- 95. Beyersmann, J., et al., A competing risks analysis of bloodstream infection after stem-cell transplantation using subdistribution hazards and cause-specific hazards. Stat Med, 2007. **26**(30): p. 5360-9.
- 96. Wolbers, M., et al., *Prognostic models with competing risks: methods and application to coronary risk prediction*. Epidemiology, 2009. **20**(4): p. 555-61.
- 97. Evans, D.W., et al., *Peritonitis-free survival in peritoneal dialysis: an update taking competing risks into account.* Nephrol Dial Transplant, 2010. **25**(7): p. 2315-22.
- 98. Coviello, E. and M.M. Boggess, *Cumulative incidence estimated in the precence of competing risks*. Stata Journal, 2002. **4**(2): p. 103-112.
- 99. Moeschberger, M.L., K.P. Tordoff, and N. Kochar, *A review of statistical analysis for competing risks*, in *Handbook of Statistics Vol.27*, C.R. Rao, J.P. Miller, and D.C. Rao, Editors. 2008, Elsevier: Amsterdam. p. 321-341.
- 100. SKAR, The Swedish Knee Arthroplasty Register Annual Report, 2008: Lund.
- Lie, S.A., et al., Mortality after total hip replacement: 0-10-year follow-up of 39,543 patients in the Norwegian Arthroplasty Register. Acta Orthop Scand, 2000. 71(1): p. 19-27.
- 102. Gillam, M.H., et al., *Different competing risks models applied to data from the Australian Orthopaedic Association National Joint Replacement Registry*. Acta Orthop, 2011. **82**(5): p. 513-20.
- 103. Aalen, O.O., A model for non-parametric regression analysis of life times., in Mathematical statistics and probability theory : proceedings, sixth international

conference, Wisla (Poland) 1978, W. Klonecki, A. Kozek, and J. Rosinski, Editors. 1980, Springer-Verlag: New York. p. 1-25.

- 104. Scheike, T.H. and M.J. Zhang, *An additive-multiplicative Cox-Aalen regression model*. Scandinavian Journal of Statistics, 2002. **29**(1): p. 75-88.
- 105. Scheike, T.H. and M.J. Zhang, *Flexible competing risks regression modeling* and goodness-of-fit. Lifetime Data Anal, 2008. **14**(4): p. 464-83.
- 106. Martinussen, T. and T.H. Scheike, *Dynamic regression models for survival data*. Statistics for biology and health. 2006, New York, N.Y.: Springer. xiii, 470 p.
- 107. Gray, R.J., *cmprsk: Subdistribution Analysis of Competing Risks*, in *R Foundation for Statistical Computing*2010.
- 108. R Development Core Team, *R: A language and environment for statistical computing*, 2011, R Foundation for Statistical Computing: Vienna, Austria.
- 109. Lau, B., S.R. Cole, and S.J. Gange, *Competing risk regression models for epidemiologic data*. Am J Epidemiol, 2009. **170**(2): p. 244-56.
- 110. Lie, S.A., et al., *Duration of the increase in early postoperative mortality after elective hip and knee replacement.* J Bone Joint Surg Am, 2010. **92**(1): p. 58-63.
- 111. AOANJRR, Australian Orthopaedic Association National Joint Replacement Registry, Annual Report, 2010, AOA: Adelaide.
- 112. Hailer, N.P., G. Garellick, and J. Karrholm, Uncemented and cemented primary total hip arthroplasty in the Swedish Hip Arthroplasty Register. Acta Orthop, 2010. **81**(1): p. 34-41.
- 113. Keating, J.F., et al., Randomized comparison of reduction and fixation, bipolar hemiarthroplasty, and total hip arthroplasty. Treatment of displaced intracapsular hip fractures in healthy older patients. J Bone Joint Surg Am, 2006. **88**(2): p. 249-60.
- 114. Frihagen, F., et al., [The treatment of femoral neck fractures]. Tidsskr Nor Laegeforen, 2010. 130(16): p. 1614-7.
- 115. Ahn, J., et al., Systematic review of cemented and uncemented hemiarthroplasty outcomes for femoral neck fractures. Clin Orthop Relat Res, 2008. **466**(10): p. 2513-8.

- 116. Parker, M.J., K.S. Gurusamy, and S. Azegami, *Arthroplasties (with and without bone cement) for proximal femoral fractures in adults.* Cochrane Database Syst Rev, 2010(6): p. CD001706.
- 117. Schemper, M. and T.L. Smith, A note on quantifying follow-up in studies of failure time. Control Clin Trials, 1996. **17**(4): p. 343-6.
- 118. Maurer, T.B., et al., *Increased loosening of cemented straight stem prostheses made from titanium alloys. An analysis and comparison with prostheses made of cobalt-chromium-nickel alloy.* Int Orthop, 2001. **25**(2): p. 77-80.
- 119. Doets, H.C., R. Brand, and R.G. Nelissen, *Total ankle arthroplasty in inflammatory joint disease with use of two mobile-bearing designs.* J Bone Joint Surg Am, 2006. **88**(6): p. 1272-84.
- 120. Graw, F., T.A. Gerds, and M. Schumacher, *On pseudo-values for regression analysis in competing risks models.* Lifetime Data Anal, 2009. **15**(2): p. 241-55.
- 121. Beyersmann, J., et al., *Simulating competing risks data in survival analysis*. Stat Med, 2009. **28**(6): p. 956-71.
- 122. Beyersmann, J. and M. Schumacher, *Misspecified regression model for the subdistribution hazard of a competing risk*. Stat Med, 2007. **26**(7): p. 1649-51.
- 123. Latouche, A., et al., *Misspecified regression model for the subdistribution hazard of a competing risk.* Stat Med, 2007. **26**(5): p. 965-74.
- 124. Grambauer, N., et al., *Incidence densities in a competing events analysis*. Am J Epidemiol, 2010. **172**(9): p. 1077-84.
- 125. Allignol, A., et al., *Understanding competing risks: a simulation point of view*. BMC Med Res Methodol, 2011. **11**: p. 86.
- 126. Dignam, J.J., Q. Zhang, and M. Kocherginsky, *The use and interpretation of competing risks regression models*. Clin Cancer Res, 2012. **18**(8): p. 2301-8.
- 127. Gillam, M.H., et al., *Multi-state models and arthroplasty histories after unilateral total hip arthroplasties*. Acta Orthop, 2012. **83**(3): p. 220-6.
- 128. de Wreede, L.C., M. Fiocco, and H. Putter, *The mstate package for estimation and prediction in non- and semi-parametric multi-state and competing risks models*. Comput Methods Programs Biomed, 2010. **99**(3): p. 261-74.
- 129. Shakoor, N., et al., *Nonrandom evolution of end-stage osteoarthritis of the lower limbs*. Arthritis Rheum, 2002. **46**(12): p. 3185-9.

- 130. Srikanth, V.K., et al., *A meta-analysis of sex differences prevalence, incidence and severity of osteoarthritis.* Osteoarthritis Cartilage, 2005. **13**(9): p. 769-81.
- 131. Cook, R.J. and P. Major, *Multistate analysis of skeletal events in patients with bone metastases*. Clin Cancer Res, 2006. **12**(20 Pt 2): p. 6264s-6269s.
- 132. Putter, H., et al., *Estimation and prediction in a multi-state model for breast cancer*. Biom J, 2006. **48**(3): p. 366-80.
- 133. Uhry, Z., et al., *Multi-state Markov models in cancer screening evaluation: a brief review and case study.* Stat Methods Med Res, 2010. **19**(5): p. 463-86.
- 134. Sommen, C., A. Alioum, and D. Commenges, *A multistate approach for estimating the incidence of human immunodeficiency virus by using HIV and AIDS French surveillance data.* Stat Med, 2009. **28**(11): p. 1554-68.
- 135. Gillam, M.H., et al., *The progression of end-stage osteoarthritis: analysis of data from the Australian and Norwegian joint replacement registries using a multi-state model.* Osteoarthritis Cartilage, 2012.
- 136. Salaffi, F., et al., *Health-related quality of life in older adults with symptomatic hip and knee osteoarthritis: a comparison with matched healthy controls.* Aging Clin Exp Res, 2005. **17**(4): p. 255-63.
- 137. Fransen, M., et al., *The epidemiology of osteoarthritis in Asia*. International Journal of Rheumatic Diseases, 2011. **14**(2): p. 113-21.
- 138. Brandt, K.D., et al., *Yet more evidence that osteoarthritis is not a cartilage disease*. Ann Rheum Dis, 2006. **65**(10): p. 1261-4.
- 139. Hunter, D.J., *Insights from imaging on the epidemiology and pathophysiology of osteoarthritis.* Radiologic Clinics of North America, 2009. **47**(4): p. 539-51.
- 140. Sandell, L.J., *Etiology of osteoarthritis: genetics and synovial joint development*. Nature Reviews. Rheumatology, 2012. **8**(2): p. 77-89.
- 141. Dougados, M., *Monitoring osteoarthritis progression and therapy*. Osteoarthritis Cartilage, 2004. **12 Suppl A**: p. S55-60.
- 142. Lohmander, L.S., et al., *Incidence of severe knee and hip osteoarthritis in relation to different measures of body mass: a population-based prospective cohort study.* Ann Rheum Dis, 2009. **68**(4): p. 490-6.
- 143.AustralianOrthopaedicAssociationNationalJointReplacementRegistry.AnnualReport.201120/92012];Availablefrom:

http://www.dmac.adelaide.edu.au/aoanjrr/documents/AnnualReports2011/AnnualReport_2011_WebVersion.pdf.

- 144. Andersen, P.K., S. Esbjerg, and T.I. Sorensen, *Multi-state models for bleeding* episodes and mortality in liver cirrhosis. Stat Med, 2000. **19**(4): p. 587-99.
- 145. Lohmander, L.S., et al., *Standardized incidence rates of total hip replacement for primary hip osteoarthritis in the 5 Nordic countries: similarities and differences.* Acta Orthop, 2006. **77**(5): p. 733-40.
- 146. Altman, R.D., et al., *Total joint replacement of hip or knee as an outcome measure for structure modifying trials in osteoarthritis*. Osteoarthritis Cartilage, 2005. **13**(1): p. 13-9.
- 147. Ackerman, I.N., et al., Variation in age and physical status prior to total knee and hip replacement surgery: a comparison of centers in Australia and Europe. Arthritis Rheum, 2009. **61**(2): p. 166-73.
- 148. Skinner, J., et al., *Racial, ethnic, and geographic disparities in rates of knee arthroplasty among Medicare patients.* N Engl J Med, 2003. **349**(14): p. 1350-9.
- 149. Gossec, L., et al., OARSI/OMERACT initiative to define states of severity and indication for joint replacement in hip and knee osteoarthritis. An OMERACT 10 Special Interest Group. J Rheumatol, 2011. **38**(8): p. 1765-9.
- 150. Umeda, N., et al., *Progression of osteoarthritis of the knee after unilateral total hip arthroplasty: minimum 10-year follow-up study.* Arch Orthop Trauma Surg, 2009. **129**(2): p. 149-54.
- 151. Chitnavis, J., et al., *End-stage coxarthrosis and gonarthrosis. Aetiology, clinical patterns and radiological features of idiopathic osteoarthritis.* Rheumatology (Oxford), 2000. **39**(6): p. 612-9.
- 152. Smith, P.C., et al., *Leadership and governance in seven developed health systems*. Health Policy, 2012. **106**(1): p. 37-49.
- 153. Gunther, K.P., et al., *Prevalence of generalised osteoarthritis in patients with advanced hip and knee osteoarthritis: the Ulm Osteoarthritis Study*. Ann Rheum Dis, 1998. **57**(12): p. 717-23.
- 154. Shakoor, N., et al., Asymmetric loading and bone mineral density at the asymptomatic knees of patients with unilateral hip osteoarthritis. Arthritis and Rheumatism, 2011. **63**(12): p. 3853-3858.

- 155. Milner, C.E., *Is gait normal after total knee arthroplasty? Systematic review of the literature.* Journal of Orthopaedic Science, 2009. **14**(1): p. 114-20.
- 156. Beaulieu, M.L., M. Lamontagne, and P.E. Beaule, *Lower limb biomechanics during gait do not return to normal following total hip arthroplasty*. Gait and Posture, 2010. **32**(2): p. 269-73.
- 157. Block, J.A. and N. Shakoor, *Lower limb osteoarthritis: biomechanical alterations and implications for therapy*. Current Opinion in Rheumatology, 2010. **22**(5): p. 544-50.
- 158. Ewen, A.M., et al., *Post-operative gait analysis in total hip replacement patients-a review of current literature and meta-analysis.* Gait and Posture, 2012. **36**(1): p. 1-6.
- 159. Foucher, K.C. and M.A. Wimmer, *Contralateral hip and knee gait biomechanics are unchanged by total hip replacement for unilateral hip osteoarthritis.* Gait and Posture, 2012. **35**(1): p. 61-5.
- 160. Konyves, A. and G.C. Bannister, *The importance of leg length discrepancy after total hip arthroplasty*. J Bone Joint Surg Br, 2005. **87**(2): p. 155-7.
- 161. Tanaka, R., et al., *Factors influencing the improvement of gait ability after total hip arthroplasty.* J Arthroplasty, 2010. **25**(6): p. 982-5.
- 162. Harvey, W.F., et al., Association of leg-length inequality with knee osteoarthritis: a cohort study. Ann Intern Med, 2010. **152**(5): p. 287-95.
- 163. Putter, H. and H.C. van Houwelingen, *Frailties in multi-state models: Are they identifiable? Do we need them?* Stat Methods Med Res, 2011.
- 164. Migliore, A., et al., Comparison of the performance of hip implants with data from different arthroplasty registers. J Bone Joint Surg Br, 2009. **91**(12): p. 1545-9.
- 165. Aalen, O.O., Armitage lecture 2010: Understanding treatment effects: the value of integrating longitudinal data and survival analysis. Stat Med, 2012. 31(18): p. 1903-17.